

Review

Automatic Fall Monitoring: A Review

Natthapon Pannurat¹, Surapa Thiemjarus^{2,*} and Ekawit Nantajeewarawat¹

- ¹ Sirindhorn International Institute of Technology, Thammasat University, Pathumthani 12121, Thailand; E-Mails: p natthapon@yahoo.com (N.P.); ekawit@siit.tu.ac.th (E.N.)
- ² National Electronics and Computer Technology Center, Pathumthani 12120, Thailand
- * Author to whom correspondence should be addressed; E-Mail: surapa.thiemjarus@nectec.or.th; Tel.: +66-2-5646-900 (ext. 2479).

Received: 8 April 2014; in revised form: 2 July 2014 / Accepted: 7 July 2014 / Published: 18 July 2014

Abstract: Falls and fall-related injuries are major incidents, especially for elderly people, which often mark the onset of major deterioration of health. More than one-third of home-dwelling people aged 65 or above and two-thirds of those in residential care fall once or more each year. Reliable fall detection, as well as prevention, is an important research topic for monitoring elderly living alone in residential or hospital units. The aim of this study is to review the existing fall detection systems and some of the key research challenges faced by the research community in this field. We categorize the existing platforms into two groups: wearable and ambient devices; the classification methods are divided into rule-based and machine learning techniques. The relative merit and potential drawbacks are discussed, and we also outline some of the outstanding research challenges that emerging new platforms need to address.

Keywords: fall monitoring; fall detection; fall prevention; wireless sensors; wearable sensors

1. Introduction

Falls represent one of the leading causes of deaths and injuries in the elderly population. According to Lord *et al.* [1], more than one-third of home-dwelling people aged 65 or above and two-thirds of those in residential care fall one or more times each year. More than two-thirds of people who have experienced a fall are prone to falling again [2]. Vellas *et al.* [3] reported that 219 out of 487 elderly subjects had experienced a fall during a two-year study period and one-third of which developed a fear

of falling after the incident. The psychological consequences often lead to decreased mobility and independence among elderly population [4]. Falls can occur on level surfaces, mostly in living rooms, bedrooms, kitchens, bathrooms, or hallways [5]. The rate of fall-related injuries is generally higher among women [6] and the medical costs increase rapidly with age [7]. Damages caused by falls include tissue injuries, lacerations, joint dislocations, bone fractures and head trauma. Carroll *et al.* [8] reported that the total direct medical costs of fall injuries among elderly people in the U.S. in 1997 were \$6.2 billion. The costs increased to \$19 billion in 2000 [7] and \$30 billion in 2010 [9]. Fall-related injury is considered one of the 20 most expensive medical conditions among community-dwelling elderly population [7]. Most elderly people are unable to get up by themselves after a fall and it was reported that, even without direct injures, half of those who experienced an extended period of lying on the floor (>1 h) died within six months after the incident [10].

Fall is defined as "an event which results in a person coming to rest inadvertently on the ground or other lower level". This definition has been used as a baseline in many fall prevention and fall-risk assessment studies [11–14], and covers most types of falls targeted by fall detection research. Variations of fall definitions from different perspectives of seniors, health care providers and research communities can be found in [15]. Thus far, there are several review papers on fall detection and prevention. Noury *et al.* [16,17] reported a short review on fall detection methods and proposed a set of protocols to evaluate fall detection algorithms. In the study, a fall is divided into four phases, *i.e.*, prefall, critical (impact), postfall and recovery phases, and fall detection algorithms are categorized based on whether they focus on "direct" detection of the critical phase or postfall phase. The critical phase, which consists of a sudden body movement towards the ground, lasts for approximately 300–500 ms.

In 2008, Yu [18] presented a survey on approaches and principles of fall detection. Based on the initial state, fall characteristics are divided into three classes, *i.e.*, fall from sleeping, fall from sitting, and fall from walking or standing. The detection methods are categorized based on device types into wearable, vision, and ambient devices. In 2009, Perry *et al.* [19] provided a brief survey on methods for real-time fall detection, categorizing them into methods that measure only acceleration, methods that use combination of sensors and methods that do not measure acceleration. It was also observed that every fall had a negative peak in the acceleration data; moreover, the acceleration change from positive to negative values and the speed of this change were important for fall detection.

In 2010, Hijaz *et al.* [20] presented another short survey on fall detection and daily activity monitoring, categorizing them into vision-based, ambient-sensor-based, and kinematic-sensor-based approaches. The relative weaknesses and strengths of each approach were discussed. In 2012, Mubashir *et al.* [21] presented another survey following the similar categorization, but with more detailed subcategories. Bai *et al.* [22] described the characteristics of a fall based on three parameters: weightlessness, impact, and overturning of the body. Weightlessness usually occurs at the beginning of a fall. During this period, the acceleration values along three axes are close to 0 g. This is followed by an impact which generates the peak signal, the value of which is usually greater than 1.8 g. Overturning of the body after a fall can be determined by the difference in acceleration signals along the three axes before the weightlessness stage and after the fall.

Recently, El-Bendary *et al.* [23] described causes and consequences of elderly falls, presented the contributions and challenges of existing fall detection techniques, and provided a review of commercial products for fall detection and prevention. In this study, existing commercial products

are categorized as wearable and hand-held solutions, movement sensing solutions, and anti-wandering solutions. None of the aforementioned survey provides details on implementation and evaluation that enable quantitative comparison across different systems. Hegde *et al.* [24] published a fall detection review classifying the technologies into four approaches, adding combined wearable and ambient sensors as another separate approach. Only eight wearable-sensor-based fall detection studies were compared in terms of sensor types, sensor placements and performance measures. Igual *et al.* [25] presented a survey of selected 66 out of 327 research publications on fall detection conducted during 2005 to 2012, categorizing them as context-aware-, wearable-accelerometer- and smartphone-based systems.

Although several studies have tried to establish a common evaluation benchmark, only certain aspects of the systems have been addressed. The purpose of this paper is to provide a comprehensive overview of fall monitoring techniques, related commercial products, existing research problems as well as future trends. We will provide an in-depth review of wearable-sensor-based fall detection experiments. The rest of the paper is structured as follows: Section 2 provides a generic overview of fall detection research with representative studies. Section 3 summarizes and compares important aspects of wearable fall detection fall detection systems. Section 4 presents a summary of commercial products related to fall detection/monitoring. Section 5 describes the future research trends. Section 6 concludes this paper.

2. Fall Detection Systems

Figure 1 depicts a typical fall detection system. The flow of real-time fall detection system starts with the detection device(s) sending motion data to a processing unit. When the algorithm captures a fall, alarm/action will be triggered. This can be in forms of alert sound (to attract attention and help from people in the vicinity), immediate intervention (e.g., inflating an airbag), or sending alarm messages to family members and/or caregivers. The information may also include time of the incident, location, direction, and status (conscious/unconscious) of the faller.



Figure 1. A system overview of a typical fall detection system.

The first fall monitoring system was developed in the early 1970s. It was designed to send out an alert message when a remote transmitter button was pressed by the user [26]. Automatic fall monitoring research, on the other hand, has been conducted since 1990s. Lord and Colvin [27] studied causes and effects of falls in elderly population in an attempt to prevent falls and proposed the use of an accelerometer for fall detection. The first prototype system was developed in 1998 by William et al. [28], using a piezoelectric shock sensor to detect the abnormal peak due to falling and a mercury tilt switch to detect the orientation of the wearer after a fall. One of the first fall detection experiments was based on video cameras. Gu [29] conducted an experiment using three Broadcast Television Systems (BTS) and illustrated that horizontal and vertical velocities can be used to distinguish falls from normal activities. In 2002, Prado et al. [30] developed a prototype of fall detection system based on two dual-axial accelerometers worn as a patch on the back of the user at the height of the sacrum. Noury [31] developed a smart fall sensor consisting of a piezoelectric accelerometer, a position tilt switch, and a vibration sensor. A simple algorithm was presented, however, the results showed that the algorithm was too sensitive. Degen et al. [32] presented a wrist-worn fall detector for elderly population. The device was easy to wear but yielded only 65% sensitivity. Sixsmith et al. [33] used an array of low-cost infrared cameras mounted on the wall. An alarm was triggered either by an excessive period of inactivity or detection of a fall. Experiments on 20 fall and 10 non-fall scenarios performed by an actor, however, showed that the system could detect only 30 percent of actual falls. In 2006, Kang et al. [34] developed a wrist-worn integrated health monitoring device consisting of a fall detector and modules for measuring single-channel electro-cardiogram (ECG), noninvasive blood pressure (NIBP), pulse oximetry (SpO2), respiration rate, and body surface temperature (BST). Nyan et al. [35] conducted fall detection experiments based on a combined use of a high speed camera and three gyroscopes embedded within a garment near the chest, trunk (below the armpit), and waist. The camera was used to study the body configuration during a fall while the angular rate was used for fall detection. Miaou et al. [36] conducted fall detection based on an omni-camera and personal information (e.g., ratio between height and width and a body mass index) stored in a database. The system actually performed lying detection and yielded an accuracy of 70% and 81% with and without personal information, respectively. Alwan et al. [37] designed a fall detection system based on floor vibration using a piezoelectric sensor. Even though 100% detection rate was reported, the fall events were simulated using anthropomorphic dummies.

In 2007, Srinivasan *et al.* [38] studied automatic fall detection based on a tri-axial accelerometer and Passive Infrared Sensors (PIRs). The wearable tri-axial accelerometer was placed on the subject's waist to capture fall events while the PIRs were mounted on the wall to provide longitudinal motion information. Motionless signals from PIR sensors were used to confirm fall events. A wearable sensor was not only placed on the subject's body, but also placed on an assistive device. Almeida *et al.* [39] presented a walking stick with a gyroscope embedded at its base for detecting fall and measuring walking pace. A fall was detected based on the magnitude of the resultant angular velocity along sideward and forward axes. The pace was derived from the summation of angular velocity between two adjacent peaks divided by the time interval between the two peaks. Warnings were given when a user walks faster than his/her normal speed. Lin *et al.* [40] performed fall detection based on an optical sensor and nine micro mercury switches embedded into a smart coat. The optical sensor placed on the left waist was used to detect a fall, while the micro mercury switches were used to identify the fall characteristics (*i.e.*, forward or backward) and user's activities (*i.e.*, lying, sitting, standing, or forward bending).

In 2008, Doukas and Maglogiannis [41] proposed a combined use of an accelerometer and a microphone placed on the foot. Based on short-time Fourier transform, it was reported that low frequency sound signals were generated with high amplitude during ground impact and could be used to detect fall events. Grassi et al. [42] combined the use of a 3D time-of-flight camera, a wearable Micro-Electro-Mechanical Systems (MEMS) accelerometer, and a microphone for fall detection. Three sensors integrated on a custom board were separately processed and evaluated with suitable algorithms. Popescu et al. [43] proposed to use an array of acoustic sensors for fall detection. In 2010, Rimminen et al. [44] presented the use of near-field imaging floor sensors for fall detection. Fall classification was performed using a two-state Markov chain and pose estimation based on Bayesian filtering. Tzeng et al. [45] used a floor pressure sensor for detecting high impact on the floor and an infrared camera to identify subject's actions. Bianchi et al. [46] developed their fall detection system based on a barometric pressure sensor and a tri-axial accelerometer placed on the waist. Based on the assumption that the atmospheric pressures between the waist and the ground are different, the experimental results showed that the sensor information can provide useful information towards fall detection. Dai et al. [47,48] introduced the use of mobile-phone-based fall detection. In 2011, Gjoreski et al. [49] performed fall detection by placing tri-axial accelerometers on the chest, waist, thigh, and ankle. They reported that sensor placement on the chest or waist was suitable for fall detection but a combination of four sensors yielded the best performance. Li et al. [50] detected falls by using accelerometers placed on different parts of the body (*i.e.*, chest, waist, wrist, thigh, ankle) combined with information derived from sensors placed on furniture (*i.e.*, bed and chair).

Recently, fall detection based on tri-axial accelerometers embedded in smart phones has become increasingly popular. Fang *et al.* [51] compared the accuracy of fall detection based on a smart phone positioned on the waist, chest and thigh and reported that the chest is the optimal placement. The advantage of using a smart phone for fall detection is that it can also be used to send out warning messages and/or track the location of a faller. Koshmak *et al.* [52] conducted a fall detection experiment on seven novice skiers by asking each of them to carry a smart phone while skiing. Heart rate and oxygen saturation were also measured during the experiment, and unexpected variations in pulse signals were observed during critical situations. A few studies [53,54], however, reported that a tri-axial accelerometer embedded in a smart phone can be of relatively low quality and thus resulted in poorer fall detection performance compared to other standard commercially available accelerometers. Kau and Chen [55] conducted a study on smart phone-based fall detection using a tri-axial accelerometer and an electronic compass.

The recent introduction of the Microsoft Kinect has drawn the interests of many researchers back to vision-based fall detection. Based on the Kinect, Stone *et al.* [56] presented a two-state fall detection technique and validated the system on a large dataset collected in 13 apartments. The dataset consists of approximately 3339 days of continuous data, comprising 454 falls (with 445 falls performed by actors and nine natural falls). Ma *et al.* [57] presented a depth-based human fall detection technique, combining two computer vision techniques, *i.e.*, shape-based and learning-based classifiers. An insightful review on Kinect-based computer vision research and applications can be found in [58]. Table 1 shows examples of the existing ambient-sensor-based fall detection experiments.

Author	Sensor Types	#Subjects	#Fall Types	#ADLs	Features	Methods	Performance
Ma <i>et al.</i> (2014) [57]	Kinect RGBD camera		1	5	Silhouette, curvature scale space	SVM, extreme learning machine	Sen = 91.15% Spe = 77.14% Acc = 86.83%
Stone <i>et al.</i> (2014) [56]	Kinect RGBD camera	9 9 16	3	3	Minimum vertical, average vertical velocity, maximum vertical acceleration, occlusion adjusted change in the number of elements on the x-y plane after projection of points below knee height, minimum frame-to-frame vertical velocity	On-ground event segmentation, ensemble decision trees	71% detection when near sensor and no occlusion 5% detection when distance >4.0 m
Li <i>et al.</i> (2012) [59]	Microphone array		20	20	Mel-frequency cepstral coefficients	ТВ	Sen = 97%, Spe = 97%, Acc = 97%
Zhang <i>et al</i> . (2012) [60]	Kinect RGBD camera	? 5	2	3	Deformation on the joint structure, subject's height	kNN, SVM	Acc = 76%-98%
Belshaw <i>et al.</i> (2011) [61]	Camera	? N/A	1	1	Silhouette, lighting, optical flow	SVM, LR, ANN	Sen = 92% Spe = 95%
Khan and Sohn (2011) [62]	Camera		4	2	N/A	HMM, LDA, KDA, RT, GPF	Acc = 95.8%
Rimminen <i>et al.</i> (2010) [44]	A matrix of near-field imaging floor sensors	?	9	7	Number of observations, longest dimension, sum of magnitude	Markov chain	Sen = 90.7% Spe = 90.6%
Tzeng <i>et al.</i> (2010) [45]	Infrared camera, floor pressure sensor		3	3	Floor pressure signal, standard deviation of vertical and horizontal projection histograms, aspect ratio	TB, RB	Sen = 93.3%-96.7% Sen = 88.9%-100% Acc = 91.1%-98.3%

 Table 1. Ambient-sensor-based fall detection experiments.

Table 1. Cont.

Author	Sensor Types	#Subjects	#Fall Types	#ADLs	Features	Methods	Performance
Nyan <i>et al.</i> (2008) [63]	Camera, retroreflective markers		5	5	Tilt angles of thigh and torso (with respect to the vertical axis) and their correlation	ТВ	N/A
Vishwakarma <i>et al.</i> (2007) [64]	Camera	? N/A	3	1	Aspect ratio, horizontal and vertical gradients, tilt angle of horizontal axis of bounding box (with respect to the vertical axis)	GMM, FST	Sen = 36%-100% Spe = 83%-100% Acc = 51%-100%
Alwan <i>et al.</i> (2006) [37]	Floor vibration sensor	N/A	3	1	Raw vibration signal	ТВ	Sen = 100%
Miaou <i>et al.</i> (2006) [36]	Camera	? 20	1	1	Silhouette, aspect ratio	RB	Sen = 78%-90% Spe = 60%-86% Acc = 70%-81%
Sixsmith <i>et al.</i> (2004) [33]	Infrared array camera		2	4	Vertical velocity	ANN	Sen = 35.7% Spe = 100%
Wu Ge (2000) [29]	Camera		3	6	Vertical and horizontal velocities	ТВ	Sen = 97.83% Spe = 100%

Notes: Sen = Sensitivity, Spe = Specificity, Acc = Accuracy, ADL = Activities of Daily Living, SVM = Support Vector Machine, TB = Threshold-Based, RB = Rule-Based, LR = Logistic Regression, RT = R-Transform, ANN = Artificial Neural Network, HMM = Hidden Markov Model, GMM = Gaussian Mixture Model, LDA = Linear Discriminant Analysis, KDA = Kernel Discriminant Analysis, GPF = Gaussian Probability Density Function, FST = Finite State Machine, kNN = k-Nearest Neighbor. *n* represents the number of sensory channels/subjects. n, Male; n, Female; n, Dummy; n, Young male/female; n, Unknown gender/age.

3. Wearable Fall Detection Systems

This section focuses on a detailed overview of different aspects of fall detection, including sensor types and placement, subject details, activities of daily living (ADLs) and fall protocols, features, classification methods, and performance evaluation.

3.1. Sensor Placement

The wearable sensors that have been used in fall detection include tilt switches, accelerometers, gyroscopes, pressure sensors, and microphones. Among these, an accelerometer is considered one of the most effective and commonly used devices. Existing fall detection studies have been conducted with different sensor positions. With a single wearable sensor, the most common placement is the wearer's waist [32,38,46,65–77]. Other positions are the wrist [50,78–82], head [83], neck [84], trunk [85–88], chest [89,90], back [91–93], shoulder [94], armpit [95], ear [96,97], thigh [98], or foot [41,99]. Multiple sensors are sometimes used to enhance the fall detection algorithms [49,50,63,80,100–105] or to search for an effective sensor placement [35,48,49,78,79,81,82,89,106,107]. The devices are usually placed both on the upper body (*i.e.*, head, neck, chest, trunk, and waist) and lower body (*i.e.*, thigh, angle, and foot). The most common position on the lower body part is the thigh, which is usually combined with the chest/trunk [49,50,78,100–102,104,107] or waist [49,50,80,103]. Some studies, however, placed sensors only on the upper body [35,79,81,107,108].





Several studies on optimal sensor placement have been conducted. Kangas *et al.* [81] studied fall detection by placing accelerometers on the subject's head, waist, and wrist. It was reported that the waist and head were efficient positions, while the wrist was not. Bourke *et al.* [107] placed sensors on the trunk and thigh, and reported the trunk as the better position. Bagnasco *et al.* [79] placed sensors on the waist, chest, and wrist, and reported the chest as the optimal position. Gjoreski *et al.* [49] compared sensors placed at the chest, waist, right ankle, and right thigh, and on the other hand, reported the waist as the optimal position. Dai *et al.* [48] conducted an experiment with smart phones placed on the

subject's chest, waist, and thigh. The waist was also reported as the optimal placement. With a similar experiment setting, Fang *et al.* [106] reported that better performance was achieved when the sensor was placed near the center of mass. The results showed that the chest was the optimal position, but placement on the waist was recommended as it was more comfortable. Figure 2 summarizes the different sensor positions used in existing fall detection experiments.

3.2. Experimental Protocols

Most fall detection algorithms are designed based on datasets containing a mixture of ADLs (including fall-like activities) and simulated falls. The summaries of ADLs and simulated fall protocols used in previous fall detection studies are given in Tables 2 and 3, respectively.

The common ADLs include standing, walking (level), walking (up/down stairs), running/jogging, jumping, sitting down on chair, getting up from chair, lying down on bed, getting up from bed, and picking up object from floor. Other ADLs used in few studies and are not listed in Table 2 include ice skating [52], answering phone [53], waving phone [55], having chest pain [62], climbing into bed [46], hand shaking [79], having headache [62], hopping [38], jumping into bed [50], rotating in chair [38], sitting on or getting up from different types of chairs (*i.e.*, armchair, kitchen chair, stool) [88,89,107], and getting in and out of bathtub [29]. Some studies also include in their protocols transition activities, such as sit-stand, stand-sit [67], sit-lie, lie-sit [72,85,90,103], stand-walk [90], and walk-turn-walk.

In addition to Table 3, other types of falls include backward to lateral fall, forward fall with forward arm protection [44], faint [62,105], fall forward with 90° turn, fall while picking up something, collapse [96], and fall vertically [71]. Aziz *et al.* [105] involved different loss of balance gestures in their fall protocols, for example, while descending from standing to sitting, while rising from sitting to standing, while turning, and while reaching. In [81], Kangas *et al.* asked subjects to perform forward and lateral falls with a simulated missing step (*i.e.*, when stepping down from a platform). Gjoreski *et al.* [49] included slow falls with an attempt to hold onto the furniture in their data collection protocol. Brown [74] divided falls into simple fall (fall end up lying) and complex fall (fall end up with a vertical posture). Chen *et al.* [65] and Wang *et al.* [97] divided postures before a fall into stand, sit-to-stand, stand-to-sit, walk, walk backward, stoop, jump, and lie on the bed. Anania *et al.* [87] divided falls into four types, *i.e.*, while resting, walking, running, and jumping.

Types of falls that are difficult to be classified are varied across different studies. Bianchi *et al.* [46] conducted a fall detection experiment with eight types of falls and eight types of ADLs and reported falls with recovery as the most difficult fall to detect. Kangas *et al.* [81] reported that among forward, backward, and sideward falls, backward fall was most difficult to detect. Tolkiehn *et al.* [76] found that out of 13 types of falls, forward falls onto the knees were the most difficult to classify.

3.3. Features

A feature extraction module plays an important role in fall detection. Its goal is to extract parameters which represent discriminative information, such as body orientation, mechanical vibration, impact, activities, and/or velocity profile, to be used as input of a classification model.

Reference	Collapse into chair	Get in/out of car seat	Get up from bed	Get up from chair	Get up from stool	Jump	Pick up object from floor	Lean backward	Lean forward	Lying	Lying down on bed	Lying left	Lying on back	Lying on stomach	Lying right	Rest against a wall	Running/jogging	Search for object on floor	Sitting	Sitting on bed	Sitting on chair	Sitting on chair (fast)	Sitting on floor	Sitting on stool	Sprinting	Squatting	Standing	Stooping/bending	Stretching	Stumble	Take a lift down	Take a lift up	Throwing oneself on back on a bed	Walking (level)	Walking downstairs	Walking upstairs	Walking with luggage
[55]						•											•		٠															•	•	•	
[75]			٠	•													•			•	•					•							•	•			
[84]										•							•				•		•				•	•						•	•	•	
[106]]																		•								•							•	i		
[69]			•	•			•													•	•													•	•	•	
[79]			•	•				•	•		•										•	•												•	i		
[49]											•							•			•	•															
[93]				•																	•													•	i		
[53]							•																											•	•	•	
[50]						•				•							•		٠			•					•			•				٠			
[98]				•													•				•				•									•	•	•	
[101]]										•										•						•							•	i		
[76]	•		٠	•		•	•				•					•					•							•			•	•			•	•	
[66]						•											٠				•						•							•			
[46]	•					•	•														•							•			•				•		
[65]						•					•						٠			•								•						•	•	•	
[48]																	•		•								•							•	i		
[83]				•							•						•				•													•	•	•	•
[85]																												•						•	i		
[80]										•									٠								•							٠			
[102]]									•									•									•									
[68]			•	•			•				•										•													٠	•	•	
[104						•					•											•								•				٠	•	•	

Table 2. Activities of Daily Living (ADL) protocols.

				1		1				1	1			1		1	1		1													1		1	1	-
Reference	Collapse into chair	Get in/out of car seat	Get up from bed	Get up from chair	Get up from stool	Jump	Pick up object from floor	Lean backward	Lean forward	Lying	Lying down on bed	Lying left	Lying on back	Lying on stomach	Lying right	Rest against a wall	Running/jogging	Search for object on floor	Sitting	Sitting on bed	Sitting on chair	Sitting on chair (fast)	Sitting on floor	Sitting on stool	Sprinting	Squatting	Standing	Stooping/bending	Stretching	Stumble	Take a lift down	Take a lift up	Throwing oneself on back on a bed	Walking (level)	Walking downstairs	Walking upstairs
[71]			•										•								•													•		
[86]				•																	•					٠								•		
[87]						•											•															<u> </u>		•		
[89]											•									•	•											<u> </u>		•		
[88]		•	•	•	•						•										•			•								<u> </u>		•		
[78]												•	•	•	•				•								•							•		
[41]																		•																•		
[81]							•																									<u> </u>		•	•	•
[103]									•																									•	•
[97]										•									•								٠					<u> </u>		•	•	•
[107]	•	•	•																•	•											 		•		
[99]																	•															 		•		
[82]							•																									 		•	•	•
[38]																			٠								٠					<u> </u>		•	•	•
[72]										•																						<u> </u>		•		
[35]			•	•			•													•	•											 		•		
[73]																			•													 		•		
[74]						•	•										•		•										•			 		•		•
[96]				1							1		•				•				•											ł		•		•

[3	17	[3	[3	[9	[9	[10	[8	[4	[7	18	18	[7	[1([6	[7	18	19	[8]	[8]	[4	[6	10 [4	17	[1([9	[5	[5	1/	14	[7	[1([6	13	18	17	[5	Defense	
21	<u>5 </u> 61	51	21 81	01	91)71	11	1]	81	81	6 01	1])41	81	71	121	21	51	31	81	71	61	61)1]	81	01	31	21	91	91)51	91	21	21	51	5]	Nelerello	9
•	•	•	•	•		•	•		•		_	•	•		•	•	•	•	•	•				•	•	•	•			•			•	•	•		Backwar	d fall (BW)
																												•									BW (imp	oact on back)
							•																٠														BW (end	up sitting)
																						•	•														BW (end	up lying)
	<u> </u>			<u> </u>		1	•			+	<u> </u>																										BW (end left/right	up lying
	-	<u> </u>					•			1	-																	•									BW (imp	act on hip)
					1		•			1	-	-				-	-	_	_				-	_						_					-		BW (rou flexion)	nd back, knee
	<u> </u>			<u> </u>		1				+	<u> </u>												•														BW (atte	smpt to get up)
						•				•	-				1						•																BW (kne	e flexion)
						•			Ľ		-										•																BW (legs	s straight)
						1				1	-												•		•												Collapse	into bed
				<u> </u>	٠			٠		+					•	•	-																			•	Fall	
•																						•				٠											Fall agai	nst wall
							•					٠		٠																		٠					Fall end	up sitting
																							•												•		Fall fron	ı chair
												٠		٠									•									٠			•		Fall fron	ı bed
																																		•			Fall on s	tairs
												_																•									Fall left (hands an	(impact on d elbowe)
	+			+			+			+	-	_				-	-	_	_				_	_				•	-	_				-	_	_	Fall to le	ft (impact on
	_																												_								hip) Fall righ	t (impact on
	_																												_								hands an Fall righ	<u>id elbows)</u> t (impact on
																						_					_						_	_			hip) Fall with	recoverv
							-+				-+										•	•	•														(then wa	lking)
																						•	•														Fall with (then sta	recovery nding)
•	•		•	•		•	•		•		•	٠		-	•	•		•	٠	•		•	-	•	•	•	-	•		٠			•	•	•		Forward	fall (FW)
																											•										FW knee	flexion
					1																							•									FW (imp and elbo	act on hands ws)
																												•									FW (imp	act on knee)
																							•														FW on k	nee
																						•	•														FW end	up lying
																						•	٠														FW (atte	empt to get up)
																						•															FW (atte fall)	empt to break
						•				•											•																FW (kne	e flexion)
					-	•			•	•	-										•																FW (legs	straight)
•			•				•				•			•	•	•			•	٠		•										•			•		Lateral f	all
		•	•	•		•			•			٠						•					•	•		•	-	•		٠			•	•			Fall left	
					ž	•			•	•	-										•																Falls left	(knee flexion)
						•			-	•	-										•																Falls left	(legs straight)
		٠	•	•		•	•		٠			٠						٠					•	٠		•		•		٠			•	•			Fall righ	t
						•				•	-										٠																Falls rig flevion)	ht (knee
					Ē	•			Ľ	•	-										•																Falls rigl straioht)	ht (legs
						<u> </u>				1				٠	1								-								٠	٠			-		Slip	
						<u> </u>				1				٠	1								-									٠					Syncope	
	<u> </u>			\square		+				+	\vdash			•	+	+			•			+	<u> </u>			1			•		•	•			+		Trip	

Table 4 provides a list of features used in the existing fall detection experiments described in Table 5. The features are extracted from an accelerometer (denoted by a), a gyroscope (denoted by ω), or a pressure sensor (denoted by p). Most of the features are extracted from acceleration signals. Mean, standard deviation, sum vector magnitude, and tilt angles are simple, yet informative, features commonly used in existing fall detection experiments. Generic equations such as mean, standard deviation and variance can be applied to any motion sensors and therefore the inputs of the equations are denoted by x.

Mean (F1) and standard deviation (F2) can be calculated along each of the three axes (x, y, and z). Means are informative for detection of static activities (e.g., lying, sitting, and standing), while standard deviations are informative for distinguishing between static and dynamic activities and for classifying dynamic activities (e.g., walking, running, and jumping). Others informative features for classification of static activities include the angle between the device and ground (F20, F26) and the angle between the device and the gravitational vector (F22, F23, F24, F29, F32, F33). The later can be represented in various forms, namely, tilt angle (F23), inclination angle (F24), orientation angle (F29), sagittal angle (F32), lateral angle (F33), and device orientation change (F27). To calculate the angle between the device and the gravitational vector, some studies used the Earth's standard acceleration due to gravity (g) [67,104] or the sum vector magnitude (F5) equal to 1g (assuming no movement in a static activity) [46,49] in their equations. During static activities, the motion signals measured from a device are quite stable, and therefore the standard deviation is lower than those of static activities. Besides standard deviation, several studies used signal magnitude area (F10) to distinguish between static and dynamic activities [46,72,90,91].

No.	Feature	Equation
F1	Mean	$\mu = \frac{1}{N} \sum_{i=1}^{N} x_i$
F2	Standard deviation	$\sigma = \sqrt{\frac{1}{N} \sum_{i=1}^{N} (x_i - \mu)^2}$
F3	Variance	$\sigma^2 = \frac{1}{N} \sum_{i=1}^{N} \left(x_i - \mu \right)^2$
F4	Standard deviation magnitude	$\left \sigma\right = \sqrt{\sigma_x^2 + \sigma_y^2 + \sigma_z^2}$
F5	Sum vector magnitude	$\left a\right = \sqrt{a_x^2 + a_y^2 + a_z^2}$
F6	Sum vector on horizontal plane	$\left a\right _{h} = \sqrt{a_{x}^{2} + a_{z}^{2}}$
F7	Standard deviation of sum vector magnitude	$\sigma_{ a } = \sqrt{\frac{1}{N} \sum_{i=1}^{N} \left(a _i - \mu_{ a } \right)}$
F8	Difference between maximum and minimum values of sum vector magnitude	$\Delta a _{\max-\min} = \max(a) - \min(a)$
F9	Root mean square of sum vector magnitude	$\left a\right _{\mathrm IMS} = \sqrt{\frac{1}{N} \sum_{i=1}^{N} \left a\right _{i}^{2}}$
F10	Signal magnitude area	$SMA = \frac{1}{t} \left(\int_{0}^{t} \left a_{x}(t) \right dt + \int_{0}^{t} \left a_{y}(t) \right dt + \int_{0}^{t} \left a_{z}(t) \right dt \right)$

Table 4. Features for fall detection experiments.

No.	Feature	Equation
F11	Activity signal magnitude area	ASMA = $\frac{1}{t_2 - t_1} \begin{pmatrix} t_2 \\ \int \\ t_1 \\ v_1 \end{pmatrix} \sqrt{a_x^2(t) + a_y^2(t) + a_z^2(t) dt}$
F12	Reference velocity	$v_{ref} = \int_{t_{rest}}^{t_{tilt}} (a(t) - g) dt$
F13	Velocity	$v = \int \left(\left a(t) \right - g \right) dt$
F14	Velocity (approximate)	$v_2 = \sqrt{\left(\int a_x(t)dt\right)^2 + \left(\int a_y(t)dt\right)^2 + \left(\int a_z(t)dt\right)^2} - \int gdt$
F15	Vertical acceleration	$a_{v} = \frac{1}{2g} \left(a ^{2} - a ^{2}_{dynamic} - g^{2} \right)$
F16	Maximum vertical acceleration	$(a_v)_{\max} = \max(a_z)$
F17	Average acceleration change	$\overline{\Delta a } = \frac{1}{T_n - T_0} \sum_{i=0}^{n-1} (a(t+1) - a(t))$
F18	Overall acceleration value	$\mathbf{a}_{overall} = \mathbf{E}\left[\left \left \mathbf{a}\right ^2 - \mathbf{E}\left[\left \mathbf{a}\right ^2\right]\right \right]$
F19	Acceleration amplitude at absolute vertical direction	$\left a_{y}\right = \left a_{x}\sin\theta_{z} + a_{y}\sin\theta_{y} - a_{z}\cos\theta_{y}\cos\theta_{z}\right $
F20	Angle between device and ground	$\rho_x = \sin(a_x), \rho_y = \sin(a_y), \rho_z = \sin(a_z)$
F21	Angle between device and gravity	$\theta_x = \sin^{-1}\left(\frac{a_x}{g}\right), \theta_y = \sin^{-1}\left(\frac{a_y}{g}\right)$
F22	Angle between z axis and vertical (with respect to the gravity)	$\theta = \operatorname{atan2}\left(\sqrt{a_x^2 + a_y^2}, a_z\right)$
F23	Tilt angle (with respect to the gravity)	$\theta = \cos^{-1}(a_z)$
F24	Inclination angle (with respect to the gravity)	$\theta = \cos^{-1} \left(\frac{a_z}{g} \right)$
F25	Posture (inclination angle with respect to the gravity, calculated using dot-product method)	$\Theta(t) = \cos^{-1} \left(\frac{\vec{g}_{s}(t) \cdot \vec{g}_{r}}{\left \vec{g}_{s}(t) \right \cdot \left \vec{g}_{r} \right } \right) \left(\frac{180}{\pi} \right)$
F26	Orientation of person's trunk (with respect to the ground)	$\rho = \tan^{-1} \left(\frac{\sqrt{a_x^2 + a_y^2}}{a_z} \right)$
F27	Device orientation change	$\theta = \cos^{-1} \left(\frac{a_x \mu_x + a_y \mu_y + a_z \mu_z}{\sqrt{\mu_x^2 + \mu_y^2 + \mu_z^2} \cdot \sqrt{a_x^2 + a_y^2 + a_z^2}} \right) \left(\frac{180}{\pi} \right)$
F28	Orientation change	$\boldsymbol{\theta} = \overline{\mathbf{a}}(t_b) \cdot \overline{\mathbf{a}}(t_a)$
F29	Orientation angle (with respect to the gravity)	$\theta = \cos^{-1} \left(\frac{a_z}{\sqrt{a_x^2 + a_y^2 + a_z^2}} \right)$
F30	Ratio between two consecutive angles	$\Theta_{ratio} = \frac{\Theta(t_i)}{\Theta(t_{i+1})}$

Table 4. Cont.

No.	Feature	Equation
F31	Difference between two consecutive angles	$\Delta \boldsymbol{\Theta} = \boldsymbol{\Theta} \left(\boldsymbol{t}_{i+1} \right) - \boldsymbol{\Theta} \left(\boldsymbol{t}_{i} \right)$
F32	Sagittal angle (with respect to the gravity)	$\theta_s = -\tan^{-1} \left(\frac{a_y}{a_z} \right) \left(\frac{180}{\pi} \right)$
F33	Lateral angle (with respect to the gravity)	$\theta_l = \tan^{-1} \left(\frac{a_x}{\sqrt{1 - a_x^2}} \right) \left(\frac{180}{\pi} \right)$
F34	Horizontal angle from <i>x</i> -axis in <i>xy</i> -plane	$\theta_h = \tan^{-1} \left(\frac{a_x}{a_y} \right)$
F35	Vertical angle from <i>x</i> -axis	$\theta_{v} = \sin^{-1} \left(\frac{\sqrt{a_{x}^{2} + a_{y}^{2}}}{ a } \right) = \cos^{-1} \left(\frac{a_{z}}{ a } \right)$
F36	Jerk (rate of acceleration change)	$\frac{\Delta a}{\Delta t} = \frac{a_x(t_i) - a_x(t_{i-1})}{0.001}$
F37	Trunk angle	$\theta_{pitch} = \int_{t=-1.2s}^{t=0.5s} \omega_{pitch}(t) dt, \theta_{roll} = \int_{t=-1.2s}^{t=0.5s} \omega_{roll}(t) dt$
F38	Trunk angular acceleration	$\alpha_{pitch} = \frac{d}{dt} \left\{ \omega_{pitch} \right\}_{-0.5s}^{0.5s}, \ \alpha_{roll} = \frac{d}{dt} \left\{ \omega_{roll} \right\}_{-0.5s}^{0.5s}$
F39	Resultant angular acceleration	$\alpha_r = \sqrt{\alpha_{pitch}^2 + \alpha_{roll}^2}$
F40	Resultant angular velocity	$\omega_{\rm r} = \sqrt{\omega_{pitch}^2 + \omega_{roll}^2}$
F41	Resultant change in trunk angle	$\theta_{\rm r} = \sqrt{\theta_{pitch}^2 + \theta_{roll}^2}$
F42	Differential pressure	$\Delta p_i = \frac{t}{2} \begin{bmatrix} k = i + (2/t) & k = i \\ \sum p_k - \sum p_k - \sum p_k \\ k = i & k = i - (t/2) \end{bmatrix}$
F43	Multiple regression equation	$Y = -0.139 + 0.0195X_1 + 0.0163X_2$
F44	Maximum acceleration derivative	N/A
F45	Maximum peak-to-peak acceleration amplitude	N/A
F46	Maximum peak-to-peak acceleration derivative	N/A
F47	Timestamp of falling body to be at rest	N/A
F48	Timestamp of body's initial contact to ground	N/A
F49	Time difference between when inclination angle exceed a critical angle and inclination velocity has local maximum above a threshold	N/A
F50	Variation of $ a $ around 1 g	N/A

Table 4. Cont.

Notes: N = number of data samples, x = observation, i = index of data sample, $g = 9.81 \text{ ms}^{-2}$, a_x , a_y , a_z , are acceleration values along the x- (sideward), y-(forward), and z- (upward) axes, respectively, $\overline{\mathbf{a}} =$ average acceleration vector, $t_a =$ time before fall, $t_b =$ time after fall, $t_{tilt} =$ time when body tilts, $t_{rest} =$ initial time when body is at rest, $\overline{g}_s =$ gravity vector estimated with respect to the body segment, $\overline{g}_r =$ the reference gravitational vector, $X_1 =$ the absolute peak value in the movement direction, $X_2 =$ the absolute peak value in the horizontal direction.

				S	ensor	rs Lo	cation	1					es				
Authors	Head	Neck	Ear	Chest	Trunk	Back	Waist	Wrist	Thigh	Angle	Foot	# Subjects	#Fall Typ	#ADLs	Features	Methods	Performance
Kau and Chen (2014) [55]									M			?	1	8	F2, F5	SVM	Sen = 92% Spe = 99.75%
Dumitrache and Pasca (2013) [75]							A(3)						5	8	F5, F50	TB, RB	Sen = 90.91%-100% Spe = 61.51%-96.70% Acc = 71.77%-95.16%
Baek et al. (2013) [84]		A(3)										?	5	9	F5, F20	TB, RB	Sen = 81.6%
Koshmak et al. (2013) [52]							R					° 7	4	1	F18, F21	TB	Sen = 100% Spe = 94%
Fang et al. (2012) [106]				M			M		M			? N/A	1	3	F5, F19	TB, RB	Sen = 72.22% Spe = 73.78%
Kangas et al. (2012) [69]							A(3)						6	8	F5, F15	TB, RB	N/A
Aziz and Robinovitch (2011) [105]	A(3)			A(3)			A(3)			A(3)			9	-	F1, F3	LDA	Acc = 97%-98%
Bagnasco et al.(2011) [79]				A(3)			A(3)	A(3)					4	9	F5	TB, RB	Sen = 71%-88% Spe = 89%-100%
Gjoreski et al. (2011) [49]				A(3)			A(3)		A(3)	A(3)		•1 7	4	4	F1, F2, F5, F9, F17, F29	TB, RB, DT, NB	Acc = 94%
Hsieh et al. (2011) [70]							A(3)						8	-	F5, F6, F12	TB, RB	Acc = 95.8%
Jacob et al. (2011) [93]						A(3) G(3)						9	4	3	F5	TB, RB	Sen = 100%
Lee et al. (2011) [53]							M						4	8	F5	TB, RB	Sen = 77%-96% Spe = 81%-82%
Li and Stankovic (2011) [50]				A(3) G(3)			A(3) G(3)	A(3) G(3)	A(3) G(3)	A(3) G(3)		Å	5	10	F1, F2, F5, F27	TB, RB, NB, k-mean	N/A
Lopes et al. (2011) [98]									M			?	3	7	N/A	TB, RB	N/A

 Table 5. Existing fall detection experiments.

				S	Senso	rs Lo	catior	1									
Authors	Head	Neck	Ear	Chest	Trunk	Back	Waist	Wrist	Thigh	Angle	Foot	# Subjects	#Fall Types	#ADLs	Features	Methods	Performance
Ojetola et al. (2011) [101]				A(3) G(3)					A(3) G(3)			ů. ×	4	4	F5	DT	Acc = 98.91%-99.45%
Tolkiehn <i>et al.</i> (2011) [76]							A(3) P						13	12	F4, F7, F30, F31	TB, RB	Acc = 81%-86.97%
Wang et al. (2011) [66]							A(3)					n 13	1	5	N/A	HMM, GMM	Acc = 94.8%
Bianchi et al. (2010) [46]							A(3) P					A B A B A A B A A B A B A B A B A B A B A B A B A B A B A B A B A B A B A B A A B A A B A B A A A A A A A A A A	8	8	F5, F10, F29, F42	TB, RB	Sen = 97.5% Spe = 96.5% Acc = 96.9%
Bourke et al. (2010) [67]							A(3)						8	4	F5, F13, F25	TB, RB	Sen = 100% Spe = 100%
Chen et al. (2010) [65]							A(3)					• 3	8	8	F5, F6, F12	TB, RB	N/A
Dai <i>et al.</i> (2010) [48]				M			M		M			1 3 2	3	4	F5, F19	TB, RB	Sen = 97.33% Spe = 92.3%
Enomoto et al. (2010) [83]	A(3) G(3)											4	4	8	F43	TB, RB	Sen = 100%
Jantaraprim <i>et al.</i> (2010) [85]					A(3)								4	6	F5	TB, RB	Sen = 100% Spe = 96.67%-100%
Klenk et al. (2010) [92]						A(3)							1	-	F3, F36	TB, RB	N/A
Lai et al. (2010) [80]		A(3)					A(3)	A(3)	A(3)			8	1	4	F5, F11, F26	TB, RB	Acc = 99.55%
Vermeiren <i>et al.</i> (2010) [102]					A(3)				A(3)			?	3	3	F5	TB, RB	Sen = 97%
Weiss et al. (2010) [91]						A(3)							1	1	F2, F5, F10, F16, F44, F45, F46	TB, RB	Sen = 85.7% Spe = 88%-90.1%

				S	Senso	rs Lo	cation	1									
Authors	Head	Neck	Ear	Chest	Trunk	Back	Waist	Wrist	Thigh	Angle	Foot	# Subjects	#Fall Types	#ADLs	Features	Methods	Performance
Dinh and Struck (2009) [77]							A(3)					? 5	4	-	F5, F29	FIS, ANN	Sen = 94%
Kangas et al. (2009) [68]							A(3)						6	8	F5, F8, F15	TB, RB	Sen = 97.5% Spe = 100%
Li et al. (2009) [104]				A(3) G(3)					A(3) G(3)			n 3	7	12	F5, F24	TB, RB	N/A
Tamura et al. (2009) [71]							A(3) G(3)						1	4	N/A	TB, RB	Acc = 93%
Tong et al. (2009) [86]					A(3)							Å	2	5	F5, F26	TB, RB	Sen = 100% Spe = 100%
Anania et al. (2008) [87]					A(3)								4	5	F22, F49	TB, RB	Sen = 98% Spe = 100%
Bourke et al. (2008) [89]				A(3)								11 10	8	5	F5	TB, RB	Sen > 90% Spe > 99%
Bourke et al. (2008) [88]					<mark>G(2)</mark>								8	8	F36, F37, F38, F39, F40	TB, RB	Sen = 100%
Chan <i>et al.</i> (2008) [78]				A(3)				A(3)	A(3)			?	4	7	F5	TB, RB	N/A
Bourke et al. (2008) [89]				A(3)								11 10	8	5	F5	TB, RB	Sen > 90% Spe > 99%
Bourke et al. (2008) [88]					<mark>G(2)</mark>								8	8	F36, F37, F38, F39, F40	TB, RB	Sen = 100%
Nyan <i>et al.</i> (2008) [103]							A(3)		A(3) G(3)				4	7	F32, F33	TB, RB	Sen = 95.2% Spe = 100%
Wang <i>et al.</i> (2008) [97]			A(3)									Å . 3 2	8	7	F5, F6, F12, F47, F48	TB, RB	Sen = 70.48%
Bourke et al. (2007) [107]					A(3)				A(3)				8	8	F5	TB, RB	Sen = 100% Spe = 67%-100%

Table 5. Cont.

				5	Senso	rs Lo	catio	1									
Authors	Head	Neck	Ear	Chest	Trunk	Back	Waist	Wrist	Thigh	Angle	Foot	# Subjects	#Fall Types	#ADLs	Features	Methods	Performance
Doukas et al. (2007) [99]											A(3)	?	1	2	N/A	SVM	Acc up to 98.2%
Jeon et al. (2007) [90]				A(3)								? 12	4	8	F5, F10	TB, RB	Acc = 96.52%
Kangas et al. (2007) [82]	A(3)						A(3)	A(3)				2	3	4	F5, F8, F15	TB, RB	Sen up to 100% Spe up to 100%
Srinivasan et al. (2007) [38]							A(3)						4	8	F5	TB, RB	Sen = 94.79%
Karantonis <i>et al.</i> (2006) [72]							A(3)					Å 4 2	3	9	F5, F10, F23	TB, RB	Acc = 95.6%
Nyan <i>et al.</i> (2006) [35]				<mark>G(1</mark>)	G(1)		G(1)						3	6	N/A	ТВ	Sen = 100% Spe = 92.5%-97.5%
Chen et al. (2005) [73]							A(3)					2	2	2	F5, F28	TB, RB	N/A
Garrett Brown (2005) [74]							A(3)					2	2	7	F5, F33, F34, F35	TB, RB	Sen = 90% Spe = 95%
Lindemann et al. (2005) [96]			A(3)										7	5	F5, F6	TB, RB	Sen = 100%
Degen et al. (2003) [32]							A(3)					? 3	3	-	F5, F13, F14	TB	Sen = 65%

Notes: ADL = activities of daily living, SVM = Support Vector Machine, TB = Threshold-based, RB = Rule-based, DT = Decision Tree, FIS = Fuzzy Inference System, ANN = Artificial Neural



Sum vector magnitude (F5) has been used in most studies for detecting abnormal signals. The highest peak during the sudden change of this value is used to indicate a fall [32,38,46,48–50,53,65,67–70,72–75,77–82,84–86,89–91,93,96,97,101,102,104,106,107]. In [85], F5 was used to indicate the free-fall state (*i.e.*, vertical speed increases linearly with time due to gravity). Using only F5 alone, however, is not sufficient for accurate fall detection since jumping at once also generates a sudden change of this feature which can trigger a false alarm. To solve this problem, the activity that follows the peak is sometimes used. For example, if the sudden change is followed by standing, then it is not a fall. Velocity during fall (F12, F13, F14), vertical acceleration (F15), or differential pressure (F42) are sometimes used in addition to F5 to achieve higher detection accuracy [32,46,65,67,70,97]. In [83], a multiple regression equation of the absolute peak acceleration values in the movement and the horizontal directions (F43) was used to detect falls.

F37–F41 are calculated using a gyroscope. Similar to F5 of an accelerometer, the resultant angular velocity (F40) can be used to identify possible falls using a gyroscope. For fall detection, resultant angular acceleration (F39) and resultant change in trunk angle (F41) were used in [88]. To derive F39 and F41, trunk angle (F37) and trunk angular acceleration (F38) are first calculated. More accurate device orientation, particularly during dynamic motions, can be obtained by combining signals from different types of motion sensors. In [55], an electronic compass was used in conjunction with a tri-axial accelerometer to estimate the orientation of a smart phone. F44–F50 are accelerometer-based features used in existing studies with no equation provided. They are referred to in Table 5 and therefore are also listed for completeness.

3.4. Classification and Evaluation

Methods for fall classification can be broadly categorized into threshold-based, rule-based and machine learning approaches. Simple thresholding, however, is not suitable to detect different types of falls. Most automatic fall detection systems rely on the rule-based approach [38,46,47,50,65,67–74,76,78–81,83–94,96–98,102–104,106]. The main concept is to construct a set of rules for detecting ADLs or falls based on thresholds of one or more features. Commonly used features include mean, standard deviation, sum of vector magnitudes, and tilt angle. In [109], information regarding user's activities before and after a fall was also used to enhance the detection accuracy.

Several fall detection systems are based on or partly based on machine learning. Examples of algorithms used in fall detection experiments include Decision Tree (DT) [49,101], Naïve Bayes (NB) [49,50], Hidden Markov Model (HMM) [62,66], Gaussian Mixture Model (GMM) [66], k-Nearest Neighbor (kNN) [110], k-mean [50], Support Vector Machine (SVM) [41,49,61,99], Fuzzy Inference System (FIS) [77], and Artificial Neural Network (ANN) [77]. Gjoreski *et al.* [49] used machine learning algorithms, such as J48, Naïve Bayes, random forest, and SVMs, in their body posture recognition module. Li *et al.* [50] applied k-mean, Naïve Bayes, entropy discretization, and regression for the same purpose. In [101], a decision tree (C4.5) constructed from sum vector magnitude and raw data was used to distinguish between falls and ADLs. Majority filtering was applied over outputs within a non-overlap window and a fall can be identified when the majority of the sample windows reach a consensus. Zhang *et al.* [110] discriminated between falls and non-fall

activities using kNN with features extracted from their proposed Non-negative Matrix Factorization (NMF) algorithm.

Analyzing relationships between falls and resulting injuries, as well as patterns and long-term trends of fall conditions, can potentially be useful for establishing efficient fall prevention strategies. Although different types of stimulated falls have been addressed in many studies, only few studies have focused on fall type classification. Tolkiehn *et al.* [76] presented a fall detection algorithm for classifying falls into three directions (forward, backward, and left/right). Hseih *et al.* [70] proposed an algorithm that can classify eight different fall types based on tri-axial acceleration. Fall directions (forward, backward, left, and right) were first determined, followed by identifying the impact parts. Hands with elbow or hip were considered as the impact parts for falling forward left and right, back of body or hip for falling backward, and hands with elbow or knees for falling forward. Aziz *et al.* [105] proposed a system for detecting the cause of a fall, namely, slips, trips, and other types of imbalance.

To evaluate the practicality of the system, the classification model should be evaluated on datasets acquired from both young and elderly subjects. Elderly people were normally asked to perform only ADLs while young subjects were asked to perform both ADLs and simulated falls. In existing studies, the age of young subjects are between 19 and 50 years, and that of elderly subjects are between 55 and 83 years. Most fall experiments rely on only a dataset performed by young healthy subjects. However, several studies included ADLs protocols performed by elderly subjects to evaluate the false positive rates of the proposed algorithms [67,68,85,88,96,107]. The analysis of fall signals acquired from elderly subjects can be found in [69,92]. In [91], both young and elderly subjects were asked to walk on a treadmill equipped with a safety harness to protect actual falls. To induce a free fall phase, each subject walked on the treadmill for 2 min without and with obstacles randomly placed on the treadmill every few seconds.

Some studies involved more than one datasets. Tamura *et al.* [71] collected three different datasets for the development of an airbag fall prevention algorithm. To investigate the appropriate time for inflating the airbag, a dataset was acquired from fourteen young subjects while performing stimulated fall protocols without wearing the airbag (Test A). To evaluate the false positive rate, another dataset was acquired from nine physiotherapists while performing only ADL protocols (Test B). To evaluate the performance of the device, four young subjects were asked to perform a simple backward fall while wearing the prototype device with the airbag (Test C). Bianchi *et al.* [46] conducted three different experiment protocols in different environments, *i.e.*, indoor, outdoor, and both indoor and outdoor. The obtained datasets were used for algorithm development and for evaluating the false negative rate and the false positive rate. To our knowledge, there exists only one available online fall database acquired using a vision-based device [111] and there is still no publicly available database to support wearable-sensor-based fall detection research.

In addition to the number of subjects, different types of falls and ADLs involved in the experiments, as well as their complexity, should also be considered in performance evaluation of an algorithm. Most studies evaluate their models using one or more of the three performance measures calculated from a confusion matrix, namely, sensitivity, specificity, and accuracy. Sensitivity reflects the ability of the system in detecting falls, specificity reflects the ability of the system in detecting ADLs, and accuracy is the overall ability of the system in detecting both falls and ADLs. Table 5 shows a summary of existing wearable-sensor-based fall detection experiments. The listed experiment attributes include

authors, sensor types and placements, subject details (number of subjects, age range and gender), number of falls and ADLs, features (with reference to Table 4), classification methods, and performance. Types of sensors and subjects are represented using graphical icons. The stars on sensor icons indicate that sensors at different positions were combinedly used at the same time. Their absence indicates that the subject wore the device (possibly with multiple sensors) at one position at a time while performing the experiment protocols. Some of these studies can also send a real-time alert when a fall is detected [38,41,46,47,49,50,65,69–77,79,83–87,89,90,96–99,101–104,106,112].

4. Fall Detection Products

There exist several fall detection products currently available in the market. Most of them are wearable devices with ease-of-use designs. The most common device placement position is the waist, but some devices are designed to be placed on the wrist, or around the neck. Most products use an acceleration sensor and lithium battery, which has a maximum lifetime of up to 2 years. They usually come with an alarm button allowing a manual call for help when a fall is not automatically detected. Table 6 illustrates some examples of existing wearable fall detection products.

5. Future Trends

As guidelines for future fall monitoring research, this section presents the trends of fall detection devices, as well as addresses the difference between simulated and real world fall conditions. To reduce the causes of deaths and to improve the quality of life in elderly population, not only fall detection but also fall-risk assessment and fall prevention should be investigated.

5.1. Devices

The number of fall studies by using camera-based and wearable devices is still increasing [25]. Using a camera seems to be a reliable approach for fall detection, a camera-based system can provide a high percentage of sensitivity and specificity [29,35]. However, the major disadvantages of this approach are the complex setup, area constraints and the lack of privacy. Wearable sensors are therefore still more popular than cameras in fall monitoring. Furthermore, most wearable sensors are now embedded in smartphones. The advantages of using a smartphone are cost effectiveness, usability in both indoor and outdoor environments, and ability to track a user using the GPS module in the phone [53].

5.2. Real World Falls

The datasets for fall detection in most existing studies were acquired from young healthy subjects both during simulated falls and ADLs. When performing simulated falls, the subjects are often instructed to fall directly onto a mattress. However, in real world conditions, fallers may try to break their falls with their hands before collapsing on the ground or may fall slowly [46]. The impact surface can also be hard. These conditions are different from simulated falls in laboratory settings. So the generalization of the methods developed needs to be carefully assessed.

Table 6. Wearable fall detection products.

Name	SENSO [113]	MySOS Mandown [114]	SensorBand [115]	Badge-iT Fall	Fall detector MCT-241MD	VitalRase [118]	Climay Fall Sensor [119]
				Detector [116]	PERS [117]	vitaibase [110]	Chinax Fan Sensor [119]
Size (mm)	65 imes 43 imes 13	$70 \times 42 \times 17$	$51\times32\times17$	35×60	$67 \times 41 \times 21$	37 × 12	$58.6 \times 42 \times 19$
Weight (g)	N/A	43	N/A	N/A	N/A	35	N/A
Sensor Type	N/A	3D accelerometer	3D accelerometer	N/A	N/A	N/A	N/A
Placement	Waist	Neck/Waist	Waist	Waist	N/A	Wrist	Neck
Battery Type	Lithium Polymer	Lithium-ion	Lithium	Lithium	Lithium	Lithium	Lithium
Battery Life	Up to 2 years	25 h	1 year	N/A	N/A	2 years	2 years
Range (m)	N/A	N/A	N/A	450	N/A	200	130–160
	• Automatic fall detection	• Automatic fall detection	• Wireless	• Detect a fall and user	• Automatic fall detection	• Automatic fall	Automatic fall detection
	• Send emergency text from	• Dual band: 900/1800 MHz	communication	lying unconsciously	• Fit with emergency button	detection	• Waterproof
	user phone to emergency	with GPRS	• Connect to the	• Fit with emergency	• Full waterproof	Emergency call	Low battery detection
	contact(s)	GPS location information	internet	button	Include neck cord with	Patient call	Compatible with climax
	• Fit with emergency button	sent via GPRS with SMS	Always online	• When standing on a	built-in safety release	• Waterproof	medical alarm panels
	• Suitable for all levels of	backup for transmission of	and connect to	table, can be used as	mechanism and belt-clip	• An in-built	Pendant style
	user mobility	alarm messages, in poor	the database	a "knock-over"	Superior transmission range	cancellation button	• Operating temperature:
	• Usable in and away from	communication	• Speech	alarm to summon	Smart anti-collision algorithm	to cancel the call if	-10 to 40 °C
Footures	home	conditions.	communication	assistance	Support multiple simultaneous	necessary	• Humidity: up to 90%
reatures	• No need for costly call	Two-way audio:	• Detect a change	Detect wandering	transmissions	• Can be set to	non-condensing
	center support	microphone is sensitive	in angle,	using radio signal	PowerCode ID factory-selected	vibrate	
	• Battery life: 2 weeks when	up to 2 m	orientation and	strength	from 16 million possible code		
	fully charged or 2 years on	LED status indication:	impact to	Detect potential	combinations		
	transmission of 1 alarm a	low battery, GPS, GSM	differentiate a fall	hypothermia (low	Visible and transmitted low		
	day	coverage and connection	from normal	temperature for	battery indication Available in		
			ADLs	prolonged time)	several optional frequencies in		
					compliance with international		
					standards		

 Table 6. Cont.

Name	Tunstall iVi Pendant [120]	Oval Fall Detector [121]	AFrame the Watch [122]	Philips Lifeline's AutoAlert	
				Pendant [123]	
Size (mm)	$58 \times 38 \times 14$	58.6 × 42 × 19	N/A	$66 \times 30 \times 17$	
Weight (g)	25	N/A	N/A	32	
Sensor Type	N/A	N/A	N/A	N/A	
Placement	Neck/Waist/Chest	Neck	Wrist	Neck	
Battery Type	Lithium	Lithium	N/A	N/A	
Battery Life	12 months	2 years	N/A	18 months	
Range (m)	50	130	N/A	N/A	
	Adjustable sensitivity with three different levels to suit individual needs and circumstances	Automatic fall detection through multiple accelerometers and sensors	Intelligent personalized alertingUnique user identification	• Automatic fall detection and call for help	
	• Ergonomic alert button enabling people with visual impairments or limited dexterity to raise a call for help	• Programmable transmission delay time of 0–9 s	 Rule engine personalized by individual Alert & trend on any combination 	WaterproofEasily accessible	
Features	 Crescendo audible alert and status LED provide user with reassurance the device has registered a fall Not-worn alert which will notify the monitoring center if the fall detector has failed to move, indicating that it may not have been worn Cancellation button enabling the wearer to cancel activations if required, reducing the number of false calls (this function can be disabled if required) Water resistant to IP67 standard enabling the fall detector to be worn in the bathroom Automatic low battery warnings Long-life, replaceable battery in easy-open activations if an analysis in an an	 Can cancel help calls or false alarms with an 8-second press of the help button Low battery detection and supervision Able to answer incoming phone calls on the home phone line Waterproof design Can be worn around the neck as a pendant Adjustable lanyard with snap closure Operating temperature: -10 to 40 °C Humidity: up to 90% non-condensing 	of factors Health monitoring, trending and alerting Ease of installation and support Software as a service Network independent Built on open standards and APIs Interoperability Preserve investment in existing systems while extending life with new capabilities Location awareness and support Personalized wander management solutions built into the system with ability to integrate to door locks 	 Neck cords are designed to break away in the event of an emergency and are easily replaced if damaged or soiled Can call for help within the range of the communicator 	

Table 6. Cont.

Name	Tunstall Fall Detector [124]	Tynetec Fall Detector [125]	Blue Alert Fall Detection Sensor [126]	CSEM Wrist Fall Detection[127]	Tynetec Wrist Worn Fall Detector [128]	onAll Chest Strap [129]	70 Degree Verso Fall Detector [130]
Size (mm)	$75 \times 53 \times 28$	$85\times 56\times 20$	N/A	N/A	$57 \times 34 \times 16$	N/A	N/A
Weight (g)	75	68	N/A	N/A	23	N/A	N/A
Sensor Type	N/A	2D accelerometer	N/A	3D accelerometer	2D accelerometer/Pressure	N/A	N/A
Placement	Waist	Waist/Neck	Waist	Wrist	Neck	Chest	Neck/Waist
Battery Type	6V Duracell PX28L	Lithium	N/A	Lithium-Polymer	N/A	N/A	N/A
Battery Life	6 months	1 year	N/A	15 days to one month	2–3 years	10 h	3 years depending on usage
Range (m)	N/A	75	N/A	N/A	N/A	N/A	100
Feature	 Triggered by change of angle and impact of a fall Wear on front or side waist Green light and two "beeps" indicate that the fall detector has been activated Fit with emergency button 	 5 user-selectable levels of sensitivity 3 event logs; impacts, pre-alarms & alarms Weekly rental Detect fall, fall recovery, stumble, and trip Fit with emergency button 	 Automatically call for help when falling Fit with emergency button 	 LCD display Re-chargeable battery Automatic fall detection Fit with emergency button 	 5 sensitivity settings Daily battery self-test & low battery reporting 	 Automatic fall detection Heart rate monitoring Activity monitoring Fall prevention Location alarms Call button Rechargeable battery 	• Detect fall when the detector is in a position of 70° or greater from vertical

Most existing fall detection systems can detect only falls, while other types of information such as direction, time-to-fall, lying fall period, location, and fall status are not yet comprehensively addressed. Such additional information should be collected and sent to hospitals or family members when falls occur. Time-to-fall is a useful piece of information when one wants to develop a fall prevention tool such as a wearable airbag [71]. Accurate localization can help an assistant to reach a faller in a timely manner. The lying fall period, the direction of fall, and the faller status can help to estimate the severity of a fall so that appropriate medical support can be dispatched without delay.

5.3. Fall Risk Assessment and Fall Prevention

Instead of detecting adverse events and reacting in a timely manner, an even better solution would be to take a proactive approach by understanding fall risk factors and developing fall prevention mechanisms. The first step of fall prevention is to understand the risk of falling. In 2000, Brauer *et al.* [131] reported the results of their investigation on the ability of various laboratory measures and clinical tests of postural balance in predicting falls in one hundred elderly female population within a period of 6 months after the tests. Postural muscle timing, movement speed, and the center of pressure motion were recorded during one or more laboratory tasks (*i.e.*, reaction-time step task, a limit of stability, and a quiet stance balance task). Four clinical tasks include the Berg Balance Scale [132], the Functional Reach Test [133], the Lateral Reach Test [134], and the Step-Up Test [135]. Based on logistic regression models, the study reported a fall prediction rate of 77%, with the sensitivity of only 51%. The results indicate that there is still plenty of room for improvement in this research area.

Examples of popular measures for fall risk assessment include STRATIFY (St. Thomas's risk assessment tool in falling elderly inpatient) [136], TUG (Time up & go) [137], Barthel index [138], and TGBA (Tinetti Gait and Balance Assessment) [139]. TUG score, for example, is the easiest and most commonly used measure. For movement evaluation, the elderly people are asked to perform a sequence of activities including stand up from a chair, walk with normal speed for 3 meters, turn around, walk back to the chair and sit down on the chair. Time to completion is recorded and if it takes longer than 14 s, the risk of falling is considered high. King *et al.* [140] illustrated the use of a miniaturized wireless e-AR (ear-worn activity recognition) in fall risk assessment and concluded that the sensor was able to detect several features from TUG assessments. Ghasemzadeh *et al.* [141] presented a physiological monitoring system based on tri-axial acceleration and EMG sensors. The acceleration and muscle activity signals were collected while subjects were standing on a half circle ball, and were processed through a statistical machine learning technique to assess postural balance. Jiang *et al.* [142] proposed a walking model for fall risk assessment based on an accelerometer placed on the subject's waist.

Recently, several studies have focused on preventing falls. Fukaya *et al.* [143] introduced the idea of launching a jacket-worn airbag which will be inflated to protect the head and buttocks based on tri-axial acceleration signals. It was designed to protect falls from height, falls from wheelchair overturn, and falls on the same level. However, the system was validated on a dummy. Tamura *et al.* [71] developed a jacket-worn airbag to protect the head, neck, hip, and thigh based on a tri-axial accelerometer and a gyroscope. Guangyi *et al.* [144] developed a belt-worn airbag using a tri-axial accelerometer and a gyroscope.

This paper presents a comprehensive review of automatic fall monitoring. In general, automatic fall monitoring can be categorized as wearable-sensor-based, ambient-sensor-based or combined-sensor-based approaches. In several studies, cameras are employed as a type of ambient sensors. Although video cameras can provide sufficient information for patient monitoring, field-of-view constraints, lighting condition, camera positioning and lack of privacy are still major limitations. Wearable sensors have advantages over ambient sensors in terms of mobility, ease of installation, coverage areas of usage and less privacy constraint. They are thus a relatively practical and cost-effective solution to support independent living for elderly people. This study provides a timeline review of the alternative technologies and particularly focuses on wearable sensor-based fall monitoring.

To establish a fair comparison across different research studies and reflect the state-of-the-art of the technology, this paper provides an in-depth review of important aspects of existing fall detection experiments, including sensor types and positions, data collection protocols, subject groups, feature extractors, classification methods, and performance measures. Apart from a detailed summary of existing research studies, we also present an overview of available commercial products and future research directions of automatic fall monitoring. Commercial products are described and compared based on size, weight, sensor type, battery, transmission range, and features. The use of mobile devices, including smart phones, as the hardware platform is becoming increasingly popular. They can be used not only for motion sensing, but also for storing profiles, communication and detecting users' locations. In addition to fall alerts, a user's location or co-location is also another important information that allows timely assistance and intervention. Although smart phones are already being carried by users on a daily basis and seem to be a promising solution for fall monitoring, there still exist many issues to be addressed, *i.e.*, sensor fixation, varied sensor quality, and battery lifetime. At the moment, dedicated-designs of wearable fall sensors with constraint positioning are still relatively better in terms of battery lifetime and achievable accuracy.

In terms of sensor placement, several studies have reported that the best accuracy can be achieved when sensors were placed near the center of mass. However, in commercial product designs, user acceptance and usability must also be considered, perhaps with a tradeoff of accuracy. Assessment of devices' accuracy and practical values is indeed difficult. This is due to the fact that there still exists a very limited amount of natural falls datasets (particularly in elderly people) and the absence of publicly available real-life datasets for benchmarking. In many real scenarios, elderly people also tend to fall slowly, which may not be captured by existing systems. Beyond automatic fall detection, there has been a recent rapid surge of interests in fall risk assessment and fall prevention, aiming towards a proactive prevention approach of elderly care. These indicate several problems yet to be addressed by the research community.

Acknowledgments

This work was supported by the Thailand Research Fund (TRF), under Royal Golden Jubilee. Program Grant No. PHD/0247/2552, and the National Research University Project, funded by Thailand Office of Higher Education Commission.

Conflicts of Interest

The authors declare no conflict of interest.

References

- 1. Lord, S.R.; Sherrington, C.; Menz, H.B.; Close, J.C.T. *Falls in Older Peolpe: Risk Factors and Strategies for Prevention*; Cambridge University Press: New York, NY, USA, 2001.
- 2. Baraff, L.J.; Della Penna, R.; Williams, N.; Sanders, A. Practice guideline for the ED management of falls in community-dwelling elderly persons. *Ann. Emerg. Med.* **1997**, *30*, 480–492.
- 3. Vellas, B.J.; Wayne, S.J.; Romeo, L.J.; Baumgartner, R.N.; Garry, P.J. Fear of falling and restriction of mobility in elderly fallers. *Age Ageing* **1997**, *26*, 189–193.
- 4. Walker, J.A.; Howland, J. Falls and fear of falling among elderly persons livling in the community: Occupational therapy interventions. *Am. J. Occup. Ther.* **1991**, *45*, 119–122.
- 5. Gill, T.; Williams, C.; Tinetti, M.E. Environmental hazards and the risk of nonsyncopal falls in the homes of community-living older persons. *Med. Care* **2000**, *38*, 1174–1183.
- 6. Stevens, J.A.; Sogolow, E.D. Gender differences for non-fatal unintentional fall related injuries among older adults. *Inj. Prev.* **2005**, *11*, 115–119.
- 7. Stevens, J.A.; Corso, P.S.; Finkelstein, E.A.; Miller, T.R. The costs of fatal and non-fatal falls among older adults. *Inj. Prev.* **2006**, *12*, 290–295.
- 8. Carroll, N.V.; Slattum, P.W.; Cox, F.M. The cost of falls among the community-dwelling elderly. *J. Manag. Care Pharm.* **2005**, *11*, 307–316.
- 9. Costs of Falls Among Older Adults. Available online: http://www.cdc.gov/ HomeandRecreationalSafety/Falls/fallcost.html (accessed on 11 November 2013).
- Wild, D.; Nayak, U.S.; Isaacs, B. How dangerous are falls in old people at home? *Br. Med. J.* 1981, 282, 266–268.
- 11. Kumar, A.; Verma, A.; Yadav, M.; Srivastava, A.K. Fall: The accidental injury in geriatric population. *J. Indian Acad. Forens. Med.* **2011**, *33*, 175–178.
- 12. Bergland, A.; Wyller, T.B. Risk factors for serious fall related injury in elderly women living at home. *Inj. Prev.* **2004**, *10*, 308–313.
- 13. Truter, I. Falls prevention in the elderly. *South Afr. Pharm. J.* **2011**, *78*, 12–17.
- Bischoff-Ferrari, H.A.; Dawson-Hughes, B.; Staehelin, H.B.; Orav, J.E.; Stuck, A.E.; Theiler, R.; Wong, J.B.; Egli, A.; Kiel, D.P.; Henschkowski, J. Fall prevention with supplemental and active forms of vitamin D: A meta-analysis of randomised controlled trials. *BMJ* 2009, *339*, doi: 10.1136/bmj.b3692.
- Zecevic, A.A.; Salmoni, A.W.; Speechley, M.; Vandervoort, A.A. Defining a fall and reasons for falling: Comparisons among the views of seniors, health care providers, and the research literature. *Gerontologist* 2006, *46*, 367–376.
- 16. Noury, N.; Rumeau, P.; Bourke, A.K.; ÓLaighin, G.; Lundy, J.E. A proposal for the classification and evaluation of fall detectors. *Biomed. Eng. Res.* **2008**, *29*, 340–349.

- Noury, N.; Fleury, A.; Rumeau, P.; Bourke, A.K.; Laighin, G.O.; Rialle, V.; Lundy, J.E. Fall detection—Principles and methods. In Proceedings of the Twenty-Ninth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Lyon, France, 22–26 August 2007; pp. 1663–1666.
- Yu, X. Approaches and principles of fall detection for elderly and patient. In Proceedings of the Tenth IEEE International Conference on e-health Networking, Applications and Services, Singapore, 7–9 July 2008; pp. 42–47.
- Perry, J.T.; Kellog, S.; Vaidya, S.M.; Jong-Hoon, Y.; Ali, H.; Sharif, H. Survey and evaluation of real-time fall detection approaches. In Proceedings of the Sixth International Symposium on High-Capacity Optical Networks and Enabling Technologies, Alexandria, Arab Republic of Egypt, 28–30 December 2009; pp. 158–164.
- Hijaz, F.; Afzal, N.; Ahmad, T.; Hasan, O. Survey of fall detection and daily activity monitoring techniques. In Proceedings of the International Conference on Information and Emerging Technologies, Karachi, Pakistan, 14–16 June 2010; pp. 1–6.
- 21. Mubashir, M.; Shao, L.; Seed, L. A survey on fall detection: Principles and approaches. *Neurocomputing* **2013**, *100*, 144–152.
- Bai, Y.-W.; Wu, S.-C.; Tsai, C.-L. Design and implementation of a fall monitor system by using a 3-axis accelerometer in a smart phone. In Proceedings of the Sixteenth IEEE International Symposium on Consumer Electronics, Harrisburg, PA, USA, 4–6 June 2012; pp. 1–6.
- 23. El-Bendary, N.; Tan, Q.; Pivot, F.C.; Lam, A. Fall detection and prevention for the elderly: A review of trends and challenges. *Int. J. Smart Sens. Intell. Syst.* **2013**, *6*, 1230–1266.
- Hegde, R.; Sudarshan, B.G.; Kumar, S.C.P.; Hariprasad, S.A.; Satyanarayana, B.S. Technical advances in fall detection system- a review. *Int. J. Comput. Sci. Mob. Comput.* 2013, 2, 152–160.
- 25. Igual, R.; Medrano, C.; Plaza, I. Challenges, issues and trends in fall detection systems. *BioMed. Eng. Online* **2013**, *12*, doi:10.1186/1475-925X-12-66.
- 26. What's new in electronics: Emergency dailer. In *Popular Science*; Bonnier Corporation: New York, NY, USA, 1975; p. 104.
- Lord, C.J.; Colvin, D.P. Falls in the elderly: Detection and assessment. In Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Orlando, FL, USA, 31 October–3 November 1991; pp. 1938–1939.
- Williams, G.; Doughty, K.; Cameron, K.; Bradley, D.A. A smart fall and activity monitor for telecare applications. In Proceedings of the Twentieth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Hong Kong, China, 29 October–1 November 1998; pp. 1151–1154.
- 29. Gu, W. Distinguishing fall activities from normal activities by velocity characteristics. *J. Biomech.* **2000**, *33*, 1497–1500.
- Prado, M.; Reina-Tosina, J.; Roa, L. Distributed intelligent architecture for falling detection and physical activity analysis in the elderly. In Proceedings of the Second Joint of Annual Conference of the Engineering in Medicine and Biology Society and Annual Fall Meeting of the Biomedical Engineering Society, Houston, TX, USA, 23–26 October 2002; pp. 1910–1911.

- 31. Noury, N. A smart sensor for the remote follow up of activity and fall detection of the elderly. In Proceedings of the Second Annual International IEEE-EMB Special Topic Conference on Microtechnologies in Medicine and Biology, Madison, WI, USA, 2–4 May 2002; pp. 314–317.
- Degen, T.; Jaeckel, H.; Rufer, M.; Wyss, S. Speedy: A fall detector in a wrist watch. In Proceedings of the Seventh IEEE International Symposium on Wearable Computers, New York, NY, USA, 21–23 October 2003; pp. 184–187.
- Sixsmith, A.; Johnson, N. A smart sensor to detect the falls of the elderly. *IEEE Pervasive Comput.* 2004, *3*, 42–47.
- 34. Kang, J.M.; Yoo, T.; Kim, H.C. A wrist-worn integrated health monitoring instrument with a tele-reporting device for telemedicine and telecare. *IEEE Trans. Instrum. Meas.* 2006, 55, 1655–1661.
- Nyan, M.N.; Tay, F.E.H.; Tan, A.W.Y.; Seah, K.H.W. Distinguishing fall activities from normal activities by angular rate characteristics and high-speed camera characterization. *Med. Eng. Phys.* 2006, *28*, 842–849.
- Miaou, S.-G.; Sung, P.-H.; Huang, C.-Y. A customized human fall detection system using omni-camera images and personal information. In Proceedings of the First Transdisciplinary Conference on Distributed Diagnosis and Home Healthcare, Arlington, VA, USA, 2–4 April 2006; pp. 39–42.
- Alwan, M.; Rajendran, P.J.; Kell, S.; Mack, D.; Dalal, S.; Wolfe, M.; Felder, R. A smart and passive floor-vibration based fall detector for elderly. In Proceedings of the Second Information and Communication Technologies, Damascus, Syria, 24–28 April 2006; pp. 1003–1007.
- Srinivasan, S.; Jun, H.; Lal, D.; Gacic, A. Towards automatic detection of falls using wireless sensors. In Proceedings of the Twenty-Ninth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Lyon, France, 22–26 August 2007; pp. 1379–1382.
- Almeida, O.; Ming, Z.; Liu, J.C. Dynamic fall detection and pace measurement in walking sticks. In Proceedings of the Joint Workshop on High Confidence Medical Devices, Software, and Systems and Medical Device Plug-and-Play Interoperability. Boston, MA, USA, 25–27 June 2007; pp. 204–206.
- 40. Lin, C.-S.; Hsu, H.C.; Lay, Y.-L.; Chiu, C.-C.; Chao, C.-S. Wearable device for real-time monitoring of human falls. *Measurement* **2007**, *40*, 831–840.
- Doukas, C.; Maglogiannis, I. Advanced patient or elder fall detection based on movement and sound data. In Proceedings of the Second International Conference on Pervasive Computing Technologies for Healthcare, Tampere, Finland, 30 January–2 Febuary 2008; pp. 103–107.
- Grassi, M.; Lombardi, A.; Rescio, G.; Malcovati, P.; Malfatti, M.; Gonzo, L.; Leone, A.; Diraco, G.; Distante, C.; Siciliano, P.; *et al.* A hardware-software framework for high-reliability people fall detection. In Proceedings of the IEEE Sensors, Lecce, Italy, 26–29 October 2008; pp. 1328–1331.
- Popescu, M.; Yun, L.; Skubic, M.; Rantz, M. An acoustic fall detector system that uses sound height information to reduce the false alarm rate. In Proceedings of the Thirtieth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Vancouver, BC, Canada, 20–25 August 2008; pp. 4628–4631.

- 44. Rimminen, H.; Lindström, J.; Linnavuo, M.; Sepponen, R. Detection of falls among the elderly by a floor sensor using the electric near field. *IEEE Trans. Inf. Technol. Biomed.* **2010**, *14*, 1475–1476.
- 45. Tzeng, H.-W.; Chen, M.-Y.; Chen, J.-Y. Design of fall detection system with floor pressure and infrared image. In Proceedings of the International Conference on System Science and Engineering, Taipei, Taiwan, 1–3 July 2010; pp. 131–135.
- 46. Bianchi, F.; Redmond, S.J.; Narayanan, M.R.; Cerutti, S.; Lovell, N.H. Barometric pressure and triaxial accelerometry-based falls event detection. *IEEE Trans. Neural Syst. Rehabil. Eng.* **2010**, *18*, 619–627.
- 47. Dai, J.; Bai, X.; Yang, Z.; Shen, D. Mobile phone-based pervasive fall detection. *Pers. Ubiquitous Comput.* **2010**, *14*, 633–643.
- Dai, J.; Bai, X.; Yang, Z.; Shen, Z.; Xuan, D. Perfalld: A pervasive fall detection system using mobile phones. In Proceedings of the Eighth IEEE International Conference on Fervasive Computing and Communications Workshops, Mannheim, Germany, 29 March–2 April 2010; pp. 292–297.
- Gjoreski, H.; Lustrek, M.; Gams, M. Accelerometer placement for posture recognition and fall detection. In Proceedings of the Seventh International Conference on Intelligent Environments, Nottingham, UK, 25–28 July 2011; pp. 47–54.
- Li, Q.; Stankovic, J.A. Grammar-based, posture- and context-cognitive detection for falls with different activity levels. In Proceedings of the Second Conference on Wireless Health, La Jolla, CA, USA, 10–13 October 2011; pp. 1–10.
- 51. Zhao, Z.; Chen, Y.; Wang, S.; Chen, Z. Fallalarm: Smart phone based fall detecting and positioning system. *Procedia Comput. Sci.* **2012**, *10*, 617–624.
- Koshmak, G.A.; Linden, M.; Loutfi, A. Evaluation of the android-based fall detection system with physiological data monitoring. In Proceedings of the Thirty-Fifth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Osaka, Japan, 3–7 July 2013; pp. 1164–1168.
- 53. Lee, R.Y.W.; Carlisle, A.J. Detection of falls using accelerometers and mobile phone technology. *Age Ageing* **2011**, *2011*, doi:10.1093/ageing/afr050.
- 54. Albert, M.V.; Kording, K.; Herrmann, M.; Jayaraman, A. Fall classification by machine learning using mobile phones. *PLoS One* **2012**, *7*, doi:10.1371/journal.pone.0036556.
- 55. Kau, L.; Chen, C. A smart phone-based pocket fall accident detection, positioning and rescue system. *IEEE J. Biomed. Health Inf.* **2014**, *PP*, doi:10.1109/JBHI.2014.2328593.
- 56. Stone, E.; Skubic, M. Fall detection in homes of older adults using the microsoft kinect. *IEEE J. Biomed. Health Inf.* **2014**, *PP*, 10.1109/JBHI.2014.2312180.
- 57. Ma, X.; Wang, H.; Xue, B.; Zhou, M.; Ji, B.; Li, Y. Depth-based human fall detection via shape features and improved extreme learning machine. *IEEE J. Biomed. Health Inf.* **2014**, doi:10.1109/JBHI.2014.2304357.
- Han, J.; Shao, L.; Xu, D.; Shotton, J. Enhanced computer vision with microsoft kinect sensor: A review. *IEEE Trans. Cybern.* 2013, 43, 1318–1334.
- 59. Li, Y.; Ho, K.C.; Popescu, M. A microphone array system for automatic fall detection. *IEEE Trans. Biomed. Eng.* **2012**, *59*, 1291–1301.

- Zhang, C.; Tian, Y.; Capezuti, E. Privacy preserving automatic fall detection for elderly using rgbd cameras. In *Computers Helping People with Special Needs*; Miesenberger, K., Karshmer, A., Penaz, P., Zagler, W., Eds.; Springer: Berlin, Germany, 2012; Volume 7382, pp. 625–633.
- Belshaw, M.; Taati, B.; Snoek, J.; Mihailidis, A. Towards a single sensor passive solution for automated fall detection. In Proceedings of the Thirty-Third Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Boston, MA, USA, 30 August–3 September 2011; pp. 1773–1776.
- 62. Khan, Z.A.; Sohn, W. Abnormal human activity recognition system based on r-transform and kernel discriminant technique for elderly home care. *IEEE Trans. Consum. Electron.* **2011**, *57*, 1843–1850.
- 63. Nyan, M.N.; Tay, F.E.H.; Mah, M.Z.E. Application of motion analysis system in pre-impact fall detection. *J. Biomech.* **2008**, *41*, 2297–2304.
- Vishwakarma, V.; Mandal, C.; Sural, S. Automatic detection of human fall in video. In *Pattern Recognition and Machine Intelligence*; Ghosh, A., De, R.K., Pal, S.K., Eds.; Springer: Berlin, Germany, 2007; Volume 4815, pp. 616–623.
- 65. Chen, G.-C.; Huang, C.-N.; Chiang, C.-Y.; Hsieh, C.-J. A reliable fall detection system based on wearable sensor and signal magnitude area for elderly residents. In Proceedings of the Eighth International Conference on Smart Homes and Health Telematics Seoul, Korea, 22–24 June 2010; pp. 267–270.
- 66. Wang, J.; Chen, R.; Sun, X.; She, M.F.H.; Wu, Y. Recognizing human daily activities from accelerometer signal. *Procedia Eng.* **2011**, *15*, 1780–1786.
- Bourke, A.K.; van de Ven, P.; Gamble, M.; O'Connor, R.; Murphy, K.; Bogan, E.; McQuade, E.; Finucane, P.; ÓLaighin, G.; Nelson, J. Evaluation of waist-mounted tri-axial accelerometer based fall-detection algorithms during scripted and continuous unscripted activities. *J. Biomech.* 2010, 43, 3051–3057.
- 68. Kangas, M.; Vikman, I.; Wiklander, J.; Lindgren, P.; Nyberg, L.; Jämsä, T. Sensitivity and specificity of fall detection in people aged 40 years and over. *Gait Posture* **2009**, *29*, 571–574.
- 69. Kangas, M.; Vikman, I.; Nyberg, L.; Korpelainen, R.; Lindblom, J.; Jämsä, T. Comparison of real-life accidental falls in older people with experimental falls in middle-aged test subjects. *Gait Posture* **2012**, *35*, 500–505.
- Hsieh, C.-J.; Chiang, C.-Y.; Huang, C.-N.; Chan, C.-T.; Hsu, S.J. Development of fall characteristics collection system for fall prevention strategies. In Proceedings of the Fifth International Conference on Bioinformatics and Biomedical Engineering, Wuhan, China, 10–12 May 2011; pp. 1–4.
- 71. Tamura, T.; Yoshimura, T.; Sekine, M.; Uchida, M.; Tanaka, O. A wearable airbag to prevent fall injuries. *IEEE Trans. Inf. Technol. Biomed.* **2009**, *13*, 910–914.
- 72. Karantonis, D.M.; Narayanan, M.R.; Mathie, M.; Lovell, N.H.; Geller, B.G. Implementation of a real-time human movement classifier using a triaxial accelerometer for ambulatory monitoring. *IEEE Trans. Inf. Technol. Biomed.* **2006**, *10*, 156–167.
- Chen, J.; Karric, K.; Chang, D.; Luk, J.; Bajcsy, R. Wearable sensors for reliable fall detection. In Proceedings of the Twenty-Seventh Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Shanghai, China, 17–18 January 2006; pp. 3551–3554.

- Brown, G. An Accelerometer Based Fall Detector: Development, Experimentation, and Analysis; Summer Undergraduate Program in Engineering Research at Berkeley; University of California: Berkeley, CA, USA, 2005.
- Dumitrache, M.; Pasca, S. Fall detection system for elderly with gsm communication and gps localization. In Proceedings of the Eighth International Symposium on Advanced Topics in Electrical Engineering, Bucharest, Romania, 23–25 May 2013; pp. 1–6.
- 76. Tolkiehn, M.; Atallah, L.; Lo, B.; Yang, G.-Z. Direction sensitive fall detection using a triaxial accelerometer and a barometric pressure sensor. In Proceedings of the Thirty-Third Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Boston, MA, USA, 30 August–3 September 2011.
- 77. Dinh, C.; Struck, M. A new real-time fall detection approach using fuzzy logic and a neural network. In Proceedings of the Sixth International Workshop on Wearable Micro and Nano Technologies for Personalized Health, Oslo, Norway, 24–26 June 2009; pp. 57–60.
- Chan, H.-L.; Chao, P.-K.; Chen, Y.-C.; Kao, W.-J. Wireless body area network for physical-activity classification and fall detection. In Proceedings of the Fifth International Summer School and Symposium on Medical Devices and Biosensors, Hong Kong, China, 1–3 June 2008; pp. 157–160.
- 79. Bagnasco, A.; Scapolla, A.M.; Spasova, V. Design, implementation and experimental evaluation of a wireless fall detector. In Proceedings of the Fourth International Symposium on Applied Sciences in Biomedical and Communication Technologies, Barcelona, Spain, 26–29 October 2011.
- 80. Lai, C.-F.; Huang, Y.-M.; Park, J.H.; Chao, H.-C. Adaptive body posture analysis for elderly-falling detection with multisensors. *IEEE Intell. Syst.* **2010**, *25*, 20–30.
- 81. Kangas, M.; Konttila, A.; Lindgren, P.; Winblad, I.; Jamsa, T. Comparison of low-complexity fall detection algorithms for body attached accelerometers. *Gait Posture* **2008**, *28*, 285–291.
- Kangas, M.; Konttila, A.; Winblad, I.; Jamsa, T. Determination of simple thresholds for accelerometry-based parameters for fall detection. In Proceedings of the Twenty-Ninth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Lyon, France, 22–26 August 2007; pp. 1367–1370.
- Enomoto, Y.; Endo, H.; Hanawa, D.; Oguchi, K. Novel fall detection method with a wearable hybrid-type sensor. In Proceedings of the Sixth World Congress of Biomechanics, Singapore, 1–6 August 2010; pp. 1331–1334.
- Baek, W.-S.; Kim, D.-M.; Bashir, F.; Pyun, J.-Y. Real life applicable fall detection system based on wireless body area network. In Proceedings of the Tenth IEEE Consumer Communications and Networking Conference, Las Vegas, NV, USA, 11–14 January 2013; pp. 62–67.
- 85. Jantaraprim, P.; Phukpattaranont, P.; Limsakul, C.; Wongkittisuksa, B. Improving the accuracy of a fall detection algorithm using free fall characteristics. In Proceedings of the International Conference on Electrical Engineering/Electronics Computer Telecommunications and Information Technology, Chiang Mai, Thailand, 19–21 May 2010; pp. 501–504.
- Tong, L.; Chen, W.; Song, Q.; Ge, Y. A research on automatic human fall detection method based on wearable inertial force information acquisition system. In Proceedings of the IEEE International Conference on Robotics and Biomimetics, Guilin, China, 19–23 December 2009; pp. 949–953.

- Anania, G.; Tognetti, A.; Carbonaro, N.; Tesconi, M.; Cutolo, F.; Zupone, G.; de Rossi, D. Development of a novel algorithm for human fall detection using wearable sensors. In Proceedings of the IEEE Sensors Conference, Lecce, Italy, 26–29 October 2008; pp. 1336–1339.
- 88. Bourke, A.K.; Lyons, G.M. A threshold-based fall-detection algorithm using a bi-axial gyroscope sensor. *Med. Eng. Phys.* **2008**, *30*, 84–90.
- Bourke, A.K.; van de Ven, P.W.J.; Chaya, A.; Olaighin, G.; Nelson, J. Design and test of a long-term fall detection system incorporated into a custom vest for the elderly. In Proceedings of the Sixteenth Irish Signals and Systems Conference, NUI Galway, UK, 18–19 June 2008; pp. 307–312.
- Jeon, A.Y.; Kim, J.H.; Kim, I.C.; Jung, J.H.; Ye, S.Y.; Ro, J.H.; Yoon, S.H.; Son, J.M.; Kim, B.C.; Shin, B.J.; *et al.* Implementation of the personal emergency response system using a 3-axial accelerometer. In Proceedings of the Sixth International Special Topic Conference on Information Technology Applications in Biomedicine, Tokyo, Japan, 8–11 November 2007; pp. 223–226.
- 91. Weiss, A.; Shimkin, I.; Giladi, N.; Hausdorff, J. Automated detection of near falls: Algorithm development and preliminary results. *BMC Res. Notes* **2010**, *3*, 1–8.
- Klenk, J.; Becker, C.; Lieken, F.; Nicolai, S.; Maetzler, W.; Altc, W.; Zijlstra, W.; Hausdorff, J.M.; van Lummel, R.C.; Chiari, L.; *et al.* Comparison of acceleration signals of simulated and real-world backward falls. *Med. Eng. Phys.* 2010, *33*, 368–373.
- 93. Jacob, J.; Nguyen, T.; Lie, D.Y.C.; Zupancic, S.; Bishara, J.; Dentino, A.; Banister, R.E. A fall detection study on the sensors placement location and a rule-based multi-thresholds algorithm using both accelerometer and gyroscopes. In Proceedings of the IEEE International Conference on Fuzzy Systems, Taipei, Taiwan, 27–30 June 2011; pp. 666–671.
- 94. Nyan, M.N.; Tay, F.E.H.; Manimaran, M.; Seah, K.H.W. Garment-based detection of falls and activities of daily living using 3-axis mems accelerometer. *J. Phys.* **2006**, *34*, 1059–1067.
- 95. Noury, N.; Herve, T.; Rialle, V.; Virone, G.; Mercier, E.; Morey, G.; Moro, A.; Porcheron, T. Monitoring behavior in home using a smart fall sensor and position sensors. In Proceedings of the First Annual International Conference On Microtechnologies in Medicine and Biology, Lyon, France, 12–14 October 2000; pp. 607–610.
- 96. Lindemann, U.; Hock, A.; Stuber, M.; Keck, W.; Becker, C. Evaluation of a fall detector based on accelerometers: A pilot study. *Med. Biol. Eng. Comput.* **2005**, *43*, 548–551.
- 97. Wang, C.-C.; Chiang, C.-Y.; Lin, P.-Y.; Chou, Y.-C.; Kuo, I.T.; Huang, C.-N.; Chan, C.-T. Development of a fall detecting system for the elderly residents. In Proceedings of the Second International Conference on Bioinformatics and Biomedical Engineering, Shanghai, China, 16–18 May 2008; pp. 1359–1362.
- 98. Lopes, I.C.; Vaidya, B.; Rodrigues, J.J.P.C. Towards an autonomous fall detection and alerting system on a mobile and pervasive environment. *Telecommun. Syst.* **2011**, *52*, 2299–2310.
- Doukas, C.; Maglogiannis, I.; Tragas, P.; Liapis, D.; Yovanof, G. Patient fall detection using support vector machines. In Proceedings of the Fourth IFIP International Conference on Artificial Intelligence Applications and Innovations, Athens, Greece, 19–21 September 2007; pp. 147–156.

- Mostarac, P.; Malaric, R.; Jurcevic, M.; Hegedus, H.; Lay-Ekuakille, A.; Vergallo, P. System for monitoring and fall detection of patients using mobile 3-axis accelerometers sensors. In Proceedings of the IEEE International Workshop on Medical Measurements and Applications, Bari, Italy, 30–31 May 2011; pp. 456–459.
- Ojetola, O.; Gaura, E.I.; Brusey, J. Fall detection with wearable sensors--safe (smart fall detection). In Proceedings of the Seventh International Conference on Intelligent Environments, Nottingham, UK, 25–28 July 2011; pp. 318–321.
- Vermeiren, D.; Weyn, M.; Ron, G. Detecting human motion: Introducing step, fall and adl algorithms. In *Electronic Healthcare*; Kostkova, P., Ed.; Springer: Berlin, Germany, 2010; pp. 62–69.
- 103. Nyan, M.N.; Tay, F.E.H.; Murugasu, E. A wearable system for pre-impact fall detection. *J. Biomech.* **2008**, *41*, 3475–3481.
- 104. Li, Q.; Stankovic, J.A.; Hanson, M.; Barth, A.; Lach, J. Accurate, fast fall detection using gyroscopes and accelerometer-derived posture information. In Proceedings of the Sixth International Workshop on Wearable and Implantable Body Sensor Networks, Berkeley, CA, USA, 3–5 June 2009; pp. 138–143.
- 105. Aziz, O.; Robinovitch, S.N. An analysis of the accuracy of wearable sensors for classifying the causes of falls in humans. *IEEE Trans. Neural Syst. Rehabil. Eng.* **2011**, *19*, 670–676.
- 106. Fang, S.-H.; Liang, Y.-C.; Chiu, K.-M. Developing a mobile phone-based fall detection system on android platform. In Proceedings of the Computing, Communications and Applications Conference, Hong Kong, China, 11–13 January 2012; pp. 143–146.
- 107. Bourke, A.K.; O'Brien, J.V.; Lyons, G.M., Evaluation of a threshold-based tri-axial accelerometer fall detection algorithm. *Gait Posture* **2007**, *26*, 194–199.
- 108. Bao, N.; Feng, C.; Kang, Y.; Xu, L.; Du, Y.; Zhang, L.; Yang, F.; Li, Q. A method with triaxial acceleration sensor for fall detection of the elderly in daily activities. In Proceedings of the Sixth International Conference on Universal Access in Human-computer Interaction: Context Diversity, Orlando, FL, USA, 9–14 July 2011; pp. 121–130.
- 109. Pannurat, N.; Theekakul, P.; Thiemjarus, S.; Nantajeewarawat, E. Toward real-time accurate fall/fall recovery detection system by incorporating activity information. In Proceedings of the IEEE-EMBS International Conference on Biomedical and Health Informatics, Hong Kong, China, 5–7 January 2012; pp. 196–199.
- 110. Zhang, T.; Wang, J.; Xu, L.; Liu, P. Using wearable sensor and nmf algorithm to realize ambulatory fall detection. In *Advances in Natural Computation*; Jiao, L., Wang, L., Gao, X., Liu, J., Wu, F., Eds.; Springer: Berlin, Germany, 2006; Volume 4222, pp. 488–491.
- 111. Fall Detection Dataset. Available online: http://le2i.cnrs.fr/Fall-detection-Dataset?lang=fr (accessed on 17 July 2014).
- 112. Lee, Y.; Kim, J.; Son, M.; Lee, M. Implementation of accelerometer sensor module and fall detection monitoring system based on sensor network. In Proceedings of the Twenty-Ninth Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Lyon, France, 22–26 August 2007; pp. 2315–2318.
- 113. Senso. Available online: https://www.senso-connect.com/our-fall (accessed on 7 November 2013).

- 114. Mysos Mandown. Available online: http://www.skyguardgroup.com/products/mysosMandown/ (accessed on 7 November 2013).
- 115. Sensorband II. Available online: http://www.caretech.com.au/products-/fall-sensor.html (accessed on 7 November 2013).
- 116. Badge-it Fall Detector. Available online: http://www.alert-it.co.uk/Products/AssistiT#Badge-iT-Fall-and-Personal-Alarm-Badge (accessed on 7 November 2013).
- 117. Fall Detector MCT-241MD Pers. Available online: http://www.visonic.com/Products/Wireless-Emergency-Response-Systems/Fall-detector-mct-241md-pers-wer (accessed on 7 November 2013).
- 118. Vitalbase. Available online: http://www.vitalbase.fr/teleassistance/ (accessed on 7 November 2013).
- 119. Climax Fall Sensor. Available online: http://climax.manufacturer.globalsources.com/si/ 6008800221031/pdtl/Fall-prevention/1046434163/Fall-Sensor.htm (accessed on 7 November 2013).
- 120. Tunstall IVI Pendant. Available online: http://americas.tunstall.com/pages/products/Tunstall-iVi (accessed on 7 November 2013).
- 121. Oval Fall Detector. Available online: http://epicalert.com/products/pendants/oval-fall-detector/ (accessed on 7 November 2013).
- 122. Aframe the Watch. Available online: http://aframedigital.com/index.php/features (accessed on 7 November 2013).
- 123. Autoalert Pendant. Available online: http://www.lifelinesys.com/content/lifeline-products/ personal-help-buttons/auto-alert-pendant (accessed on 7 November 2013).
- 124. Tunstall Fall Detector. Available online: http://www.tunstallamerica.com/ 3_6Fall_detection_systems.htm (accessed on 7 November 2013).
- 125. Tynetec Fall Detector. Available online: http://www.tynetec.co.uk/products/products/ telecare-sensors/fall-detector (accessed on 7 November 2013).
- 126. Blue Alert Fall Detection Sensor. Available online: http://www.bluealertalarm.com/ index.cfm?page=equipment (accessed on 7 November 2013).
- 127. Csem Wrist Fall Detection. Available online: http://www.csem.ch/docs/Show.aspx/6026/ docname/Fall%20detection%20sensor%20system%20-%20WFDSS%20-%20Paper%202009.pdf (accessed on 7 November 2013).
- 128. Tynetec Wrist Worn Fall Detector. Available online: http://www.tynetec.co.uk/products/ products/telecare-sensors/wrist-worn-fall-detector (accessed on 7 November 2013).
- onAll Chest Strap. Available online: http://oncaring.com/products.php?product=onall§ion= features (accessed on 7 November 2013).
- 130. Verso 70 Degree Fall Detector. Available online: http://www.chubbcommunitycare.co.uk/files/ product_specs/Chubb_Verso_70_degree_Fall_Detector.pdf (accessed on 7 November 2013).
- Brauer, S.G.; Burns, Y.R.; Galley, P. A prospective study of laboratory and clinical measures of postural stability to predict community-dwelling fallers. J. Gerontol. A Biol. Sci. Med. Sci. 2000, 55, M469–M476.
- 132. Berg, K. Measuring balance in the elderly: Preliminary development of an instrument. *Physiother. Can.* **1989**, *41*, 304–311.
- 133. Duncan, P.W.; Weiner, D.K.; Chandler, J.; Studenski, S. Functional reach: A new clinical measure of balance. *J. Gerontol.* **1990**, *45*, M192–M197.

- 134. Brauer, S.; Burns, Y.; Galley, P. Lateral reach: A clinical measure of medio-lateral postural stability. *Physiother. Res. Int.* **1999**, *4*, 81–88.
- 135. Hill, K.D. A new test of dynamic standing balance for stroke patients: Reliability, validity and comparison with healthy elderly. *Physiother. Can.* **1996**, *48*, 257–262.
- 136. Oliver, D.; Britton, M.; Seed, P.; Martin, F.C.; Hopper, A.H. Development and evaluation of evidence based risk assessment tool (stratify) to predict which elderly inpatients will fall: Case-control and cohort studies. *Br. Med. J.* **1997**, *315*, 1049–1053.
- 137. Podsiadlo, D.; Richardson, S. The time "up & go": A test of basic functional mobility or frail elderly persons. *J. Am. Geriatr. Soc.* **1991**, *39*, 142–148.
- 138. Mahoney, F.I.; Barthel, D.W., Functional evaluation: The barthel index. *Md. State Med. J.* **1965**, *14*, 61–65.
- 139. Vaught, S.L. Gait, balance, and fall prevention. Ochsner J. 2001, 3, 94–97.
- King, R.C.; Atallah, L.; Wong, C.; Miskelly, F.; Yang, G.Z. Elderly risk assessment of falls with bsn. In Proceedings of the International Conference on Body Sensor Networks, Singapore, 7–9 June 2010; pp. 30–35.
- Ghasemzadeh, H.; Jafari, R.; Prabhakaran, B. A body sensor network with electromyogram and inertial sensors: Multimodal interpretation of muscular activities. *IEEE Trans. Inf. Technol. Biomed.* 2010, 14, 198–206.
- 142. Jiang, S.; Zhang, B.; Wei, D. Fall risk assessment and prediction based on gait analysis. In Proceedings of the Eleventh IEEE International Conference on Computer and Information Technology, Pafos, Cyprus, 31 August–2 September 2011; pp. 176–180.
- 143. Fukaya, K.; Uchida, M. Protection against impact with the ground using wearable airbags. *Ind. Health* **2008**, *46*, 59–65.
- 144. Shi, G.; Chan, C.S.; Li, W.G.; Leung, K.-S.; Zou, Y.; Jin, Y. Mobile human airbag system for fall protection using mems sensors and embedded svm classifier. *IEEE Sens. J.* **2009**, *9*, 495–503.

© 2014 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution license (http://creativecommons.org/licenses/by/3.0/).